UC Irvine UC Irvine Previously Published Works

Title

Multiple Organ System Defects and Transcriptional Dysregulation in the Nipbl+/– Mouse, a Model of Cornelia de Lange Syndrome

Permalink https://escholarship.org/uc/item/6pr5q65k

Journal PLOS Genetics, 5(9)

ISSN 1553-7390

Authors

Kawauchi, Shimako Calof, Anne L Santos, Rosaysela <u>et al.</u>

Publication Date

2009-09-01

DOI

10.1371/journal.pgen.1000650

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at https://creativecommons.org/licenses/by/4.0/

Peer reviewed

Multiple Organ System Defects and Transcriptional Dysregulation in the *Nipbl*^{+/-} Mouse, a Model of Cornelia de Lange Syndrome

Shimako Kawauchi^{1,2,3,9}, Anne L. Calof^{1,2,3,4,9,*}, Rosaysela Santos^{1,2}, Martha E. Lopez-Burks^{2,5}, Clint M. Young^{1,2}, Michelle P. Hoang^{2,5}, Abigail Chua^{2,5}, Taotao Lao⁶, Mark S. Lechner⁶, Jeremy A. Daniel⁷, Andre Nussenzweig⁷, Leonard Kitzes^{1,3}, Kyoko Yokomori⁸, Benedikt Hallgrimsson⁹, Arthur D. Lander^{2,4,5,*}

1 Department of Anatomy and Neurobiology, University of California Irvine, Irvine, California, United States of America, 2 Developmental Biology Center, University of California Irvine, Irvine, California, United States of America, 3 Center for Hearing Research, University of California Irvine, Irvine, California, United States of America, 4 Center for Complex Biological Systems, University of California Irvine, Irvine, California, University of California Irvine, Irvine, California, United States of America, 5 Department of Developmental and Cell Biology, University of California Irvine, Irvine, California, University of California, University of California, University, Philadelphia, Pennsylvania, United States of America, 7 National Cancer Institute, National Institutes of Health, Bethesda, Maryland, United States of America, 8 Department of Biological Chemistry, University of California Irvine, Irvine, California, University of California, University of California, Viniversity, Philadelphia, Pennsylvania, United States of America, 7 National Cancer Institute, National Institutes of Health, Bethesda, Maryland, United States of America, 8 Department of Biological Chemistry, University of California Irvine, Irvine, California, University of California, University of Calgary, Alberta, Canada

Abstract

Cornelia de Lange Syndrome (CdLS) is a multi-organ system birth defects disorder linked, in at least half of cases, to heterozygous mutations in the *NIPBL* gene. In animals and fungi, orthologs of *NIPBL* regulate cohesin, a complex of proteins that is essential for chromosome cohesion and is also implicated in DNA repair and transcriptional regulation. Mice heterozygous for a gene-trap mutation in *Nipbl* were produced and exhibited defects characteristic of CdLS, including small size, craniofacial anomalies, microbrachycephaly, heart defects, hearing abnormalities, delayed bone maturation, reduced body fat, behavioral disturbances, and high mortality (75–80%) during the first weeks of life. These phenotypes arose despite a decrease in *Nipbl* transcript levels of only ~30%, implying extreme sensitivity of development to small changes in *Nipbl* activity. Gene expression profiling demonstrated that *Nipbl* deficiency leads to modest but significant transcriptional dysregulation of many genes. Expression changes at the protocadherin beta (*Pcdhb*) locus, as well as at other loci, support the view that NIPBL influences long-range chromosomal regulatory interactions. In addition, evidence is presented that reduced expression of genes involved in adipogenic differentiation may underlie the low amounts of body fat observed both in *Nipbl+/-* mice and in individuals with CdLS.

Citation: Kawauchi S, Calof AL, Santos R, Lopez-Burks ME, Young CM, et al. (2009) Multiple Organ System Defects and Transcriptional Dysregulation in the *Nipbl*^{+/-} Mouse, a Model of Cornelia de Lange Syndrome. PLoS Genet 5(9): e1000650. doi:10.1371/journal.pgen.1000650

Editor: Wendy A. Bickmore, Medical Research Council Human Genetics Unit, United Kingdom

Received May 6, 2009; Accepted August 16, 2009; Published September 18, 2009

This is an open-access article distributed under the terms of the Creative Commons Public Domain declaration which stipulates that, once placed in the public domain, this work may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose.

Funding: This work was supported by grants from the NIH (P01HD052860 to ADL and ALC; R21HD050538 to MSL); the Cornelia de Lange Syndrome Foundation; and the Center for Hearing Research at the University of California Irvine. RS was supported by the University of California Irvine NIH Minority Science Program (GM 55245). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: alcalof@uci.edu (ALC); adlander@uci.edu (ADL)

These authors contributed equally to the work

Introduction

Cornelia de Lange Syndrome (CdLS; OMIM#122470) is characterized by developmental abnormalities of the cardiopulmonary, gastrointestinal, skeletal, craniofacial, neurological, and genitourinary systems [1–3]. The clinical presentation ranges from subtle dysmorphology to conditions incompatible with postnatal life. Common structural birth defects observed in CdLS include upper limb reduction (significant in just under half of cases), cardiac abnormalities (especially atrial and ventricular septal defects), and craniofacial dysmorphia (including dental and middle ear abnormalities, occasional clefting of the palate, and highly characteristic facies) [2–8]. Other findings include small head size, lean body habitus, hirsutism, ophthalmologic abnormalities, preand postnatal growth retardation, and structural abnormalities of the gastrointestinal tract (duodenal atresia, annular pancreas, small bowel duplications) [2,3,9–11]. Physiological disturbances in CdLS include moderate to severe mental retardation [12] often accompanied by autistic behaviors [13], and severe gastrointestinal reflux [14]. Although prevalence has been estimated at between $\sim 1/10,000$ and 1/50,000 births [8,15], wide phenotypic variability in the syndrome makes it likely that large numbers of mildlyaffected individuals are not being counted.

A genetic basis for CdLS was uncovered in 2004 with the demonstration that many affected individuals carry mutations in *Nipped-B-like (NIPBL*), so named for its homology to the *Drosophila* gene, *Nipped-B* [16,17]. Heterozygous *NIPBL* mutations are found in about 50% of individuals with CdLS [18]. As many of these mutations are expected to produce absent or truncated protein, haploinsufficiency is the presumed genetic mechanism [19].

NIPBL/Nipped-B protein is found in the nuclei of all eukaryotic cells, where it interacts with cohesin, the protein complex that mediates sister chromatid cohesion [20,21]. The NIPBL ortholog in fungi plays a role in loading cohesin onto chromosomes, and a

Author Summary

Cornelia de Lange Syndrome (CdLS) is a genetic disease marked by growth retardation, cognitive and neurological problems, and structural defects in many organ systems. The majority of CdLS cases are due to mutation of one copy of the Nipped B-like (NIPBL) gene, the product of which regulates a complex of chromosomal proteins called cohesin. How reduction of NIPBL function gives rise to pervasive developmental defects in CdLS is not understood, so a model of CdLS was developed by generating mice that carry one null allele of Nipbl. Developmental defects in these mice show remarkable similarity to those observed in individuals with CdLS, including small stature, craniofacial abnormalities, reduced body fat, behavioral disturbances, and high perinatal mortality. Molecular analysis of tissues and cells from Nipbl mutant mice provide the first evidence that the major role of Nipbl in the etiology of CdLS is to exert modest, but significant, effects on the expression of diverse sets of genes, some of which are located in characteristic arrangements along the DNA. Among affected genes is a set involved in the development of adipocytes, the cells that make and accumulate body fat, potentially explaining reductions in body fat accumulation commonly observed in individuals with CdLS.

role in unloading has been suggested as well. The fact that a minority of cases of mild CdLS result from mutations in the SMC1L1/SMC1A (~5%; OMIM 300590) and SMC3 (1 case; OMIM 610579) genes, which encode two of the four cohesin structural components, supports the view that CdLS is caused by abnormal cohesin function [22,23]. Consistent with the hypothesis that cohesin plays important roles during embryonic development, it was found that mutations in the cohesin regulatory protein ESCO2 cause Roberts'-SC phocomelia syndrome, another multiorgan systems birth defects syndrome [24,25]. In mice, deletion of the cohesin regulators PDS5A and PDS5B also produces a wide variety of developmental defects, some of which overlap with CdLS [26,27]. In addition, there has recently been a report of one family showing atypical inheritance of CdLS, in which both affected and unaffected siblings harbor a missense mutation in the PDS5B gene, raising the possibility of some genetic association between PDS5B and CdLS [27].

How alterations in cohesin function give rise to pervasive developmental abnormalities is largely unknown. Cohesin is involved in sister chromatid cohesion and DNA repair in many organisms, but observed alterations in cohesion and repair in individuals with CdLS are mild at best [28,29]. More recently, observations in model organisms and cultured cells have suggested that cohesin plays important roles in the control of transcription [reviewed in 18]. In *Drosophila*, for example, changes in levels of Nipped-B or cohesin structural components alter the expression of developmental regulator genes, such as homeodomain transcription factors [30–33]. Such effects on gene expression, which have been proposed to reflect the disruption of long-range promoter-enhancer communication, occur with small changes in Nipped-B or cohesin levels that do not produce cohesion defects; they can also occur in postmitotic cells, in which chromosome segregation is presumably not an issue [18].

Studies using *Drosophila* cell lines have demonstrated that cohesin and Nipped-B binding are concentrated near the promoters of active transcriptional units [34]. In mammalian cells, cohesin often binds, in an NIPBL-dependent manner, to sites occupied by the transcriptional insulator protein CTCF, where it plays a significant role in CTCF function [35–37]. Recently, NIPBL has also been shown to bind and recruit histone deacetylases to chromatin [38]. These observations suggest that cohesin and NIPBL may interact in multiple ways with the transcriptional machinery.

As a first step toward understanding the molecular etiology of CdLS, we generated a mouse model of *Nipbl* haploinsufficiency, which replicates a remarkable number of the pathological features of CdLS. Cellular and molecular analysis of mutant cells and tissues revealed widespread, yet subtle, changes in the expression of genes, some of which are found in genomic locales in which transcription is known to be controlled through long-range chromosomal interactions. We propose that the aggregate effects of many small transcriptional changes are the cause of developmental abnormalities of CdLS, and present evidence that one set of transcriptional changes may explain the notably lean body habitus of many individuals with CdLS.

Results

Heterozygous mutation of *Nipbl* is associated with perinatal mortality

Two mouse ES cell lines bearing gene-trap insertions into Nipbl were obtained and injected into C57BL/6 blastocysts to produce chimeras (see Materials and Methods). Male chimeras were bred against both inbred (C57BL/6) and outbred (CD-1) mice. For only one cell line (RRS564, which contains a beta-geo insertion in intron1, and is predicted to produce a truncated transcript with no open reading frame; Figure S1) was ES cell contribution to the germline obtained (as scored by coat color; Table 1). Whereas Mendelian inheritance predicts that half the germline progeny of chimeric mice should be heterozygous (Nipbl+/-) for the gene trap insertion, the observed frequency was much lower. When chimeras were bred against CD-1 females, 22 out of 113 germline progeny (19%) carried the mutant allele (Table 1). With C57BL/6 females only one out of 18 germline progeny carried the mutation (5.5%), and this animal, although male, did not produce any progeny when subsequently mated.

In view of these data, it was decided that further analysis of the Nipbl- allele would take place through outcrossing onto the CD-1 background. As shown in Table 1, when the Nipbl+/- offspring of chimera by CD-1 crosses (N₀ generation) were bred against wildtype (CD-1) females, 17% of surviving adult progeny carried the mutant allele. When animals of this "N₁ generation" were again outcrossed against CD-1, 18% of surviving progeny (N₂) carried the mutant allele. Similar survival ratios were observed for subsequent generations of outcrossing.

The data imply that 75–80% of Nipbl+/- mice die prior to genotyping (typically done at 4 weeks of age), a fraction that remains stable as the mutant allele is progressively outcrossed onto the CD-1 background. To determine whether lethality occurs in utero, we examined litters for Nipbl+/- embryos just before birth (gestational days E17.5 and E18.5). With no visible marker available for the ES-cell derived progeny of chimeras, this test was carried out with progeny of the N₀ generation, in which the Mendelian expectation for the mutant allele is 50%. Mutants were found to comprise 41% (30 out of 67) of progeny, a frequency not significantly different from expected for this sample size (Table 1). These data imply that most mutants die at or after birth.

Abnormal heart and bone development, and reduced growth in *Nipbl*+/- embryos

To evaluate the extent to which Nipbl+/- mice provide a good model for CdLS, we performed an analysis in which we examined these animals for a number of different structural phenotypes analogous to common clinical findings observed in CdLS

Table 1. Heterozygosity for Nipbl causes postnatal lethality.

Paternal genotype	Surviving≥3 postnatal weeks			Viable at E17.5-E18.5			Resorption at E17.5	
	+/+	+/-	ratio	+/+	+/-	ratio	+/+	+/-
Chimera	91	22	4.1:1*	nd	nd	nd	nd	nd
N ₀ Nipbl+/-	190	40	4.8:1*	37	30	1.2:1†	0	2
N ₁ Nipbl+/-	247	54	4.6:1*	nd	nd	nd	nd	nd

Data are presented on the frequencies of genotypes resulting from crosses between heterozygous mutant males and CD-1 females. The males referred to in the first row were the chimeras produced by injection of Nipbl+/- ES cells into C57BL/6 blastocysts, so in this case only progeny descended from ES cells (as distinguished by chinchilla coat color) were scored. Surviving Nipbl+/- mice from these crosses are referred to as the No generation; their offspring with CD-1 females are referred to as the N1 generation; their offspring with CD-1 females as the N2 generation; and so on. Note that although the ratio of mutant to wildtype animals at E17.5-E18.5 is not significantly different from 1:1, the presence of identifiably-mutant, but not wildtype, resorbed embryos at this stage suggests that there may be a small amount of late embryonic loss.

 $^{*}P < 0.001$ by chi-squared analysis when compared with Mendelian expectations.

 $^{\dagger}P = 0.67$ compared with Mendelian expectations, and P<0.005 when compared with the postnatal distribution.

doi:10.1371/journal.pgen.1000650.t001

(summarized in Table S1). Among the most common clinical features of CdLS are small body size, often evident before birth; heart defects; and upper limb abnormalities ranging from small hands to frank limb truncations [2-5,7,8,39]. As shown in Table 2, Nipbl+/- embryos examined shortly before birth (E17.5-E18.5) were 18-19% smaller than wildtype littermates (P<0.001), a reduction not accompanied by decreased placental size. Nipbl+/embryos at earlier stages were also noted to be slightly smaller than littermates (data not shown).

Nipbl+/- embryos did not display limb or digit truncations, or obvious loss of any other bony elements. However, upon staining embryonic skeletons, we observed delays in ossification of both endochondral and membranous bones of Nipbl+/- embryos. As shown in Figure 1A-1D, delayed ossification of the skull and digits was apparent between E16.5 and E18.5. Measurement of long bones and digits at E17.5 revealed, in addition to a symmetrical reduction in bone length (consistent with smaller body size), a significant decrease in the relative extent of ossification (Figure 1E). Otherwise, the patterning of cartilaginous elements was relatively normal, although some subtle differences in morphology were consistently observed, e.g. the shape of the olecranon process of the ulna was consistently abnormal in Nipbl+/- mouse embryos (Figure 1F-1G). Interestingly, dys- and hypoplastic changes of the ulna are common findings in CdLS [40].

Among the cardiac defects that occur in CdLS, atrial and ventricular septal defects are especially common [2,5,7]. Atrial septal defects, which were typically large, were observed in about half of Nipbl+/- mouse embryos, (Figure 1H-1K; Table S1), and could be detected as early as E15.5, shortly after atrial septation normally finishes. A reduction in atrial size was also seen in some mutants, but was not a consistent finding. No defects were detected in the atrioventricular valves or septum, outflow tract, or pulmonary vasculature. However, many mutant embryos displayed subtle abnormalities of the ventricular and interventricular myocardium, including abnormal lacunar structures and disorganization of the compact layer, especially near the apex (data not shown). Significantly, no histological or functional cardiac abnormalities were detected among mutant mice that survived the perinatal period (data not shown). This implies that the cause of perinatal mortality is either cardiac, or correlates strongly with the presence of cardiac structural defects.

Histological examination of other organ systems in late embryonic mutant mice revealed no obvious anatomical abnormalities of the lungs, diaphragm, liver, stomach, spleen, kidney or bladder. Brains of neonatal Nipbl+/- mice displayed relatively normal gross anatomy, although a single mutant was observed to have a large brainstem epidermoid cyst (not shown).

Surviving mutant mice are small, and fail to thrive

Most Nipbl+/- mice that survived the perinatal period reached adulthood, and appeared to have a normal lifespan. However, marked decrease in the body size of mutant mice was evident at birth and throughout all ages (Figure 2A and 2B). Indeed, the 18-19% weight difference between mutant and wildtype mice observed before birth (Table 2) widens to 40-50% by postnatal weeks 3-4 (Figure 2C-2E; this finding has remained consistent over 6 generations [data not shown]). To investigate early postnatal growth of Nipbl+/- mice in more detail, litters fathered by N_1 and N_2 generation animals were subjected to daily weighing from shortly after birth until sexual maturity (5-6 weeks of age; Figure 2F). Most mutant mice exhibited failure to thrive during the first weeks of life, with many undergoing several days of wasting

Table 2. Prenatal growth retardation in *Nipbl*+/- embryos.

Genotype	E17.5 Bo	E17.5 Body weight (g)			E17.5 Placental weight (g)			E18.5 Body weight (g)			
	Mean	S.D.	Ν	Mean	S.D.	Ν	Mean	S.D.	N		
+/+	1.02	0.096	30	0.115	0.015	23	1.27	0.076	7		
Nipbl+/-	0.80*	0.115	23	0.112†	0.024	17	1.00*	0.082	7		

Embryos were obtained at the indicated times from crosses of No Nipbl+/- males and CD-1 females. Reduction in weight of 18–19% in mutant embryos is apparent at both ages, without a significant difference in placental weight.

*P<0.001. P = 0.64, by Student's t-test.



Figure 1. Abnormalities of bone and heart development in Nipbl+/- mice. (A,B) Cranial and trunk skeleton at E17.5, stained with alcian blue/alizarin red (blue staining shows cartilage; red shows bone). (A) Wildtype and (B) Nipbl+/-. Arrows indicate locations at which ossification is delayed in the mutant. (C) Forepaws at E16.5. Skeletal elements are patterned normally in the mutant, but are smaller. (D) Forepaws at E18.5. Delayed ossification is readily seen in mutant metacarpals and phalanges. Scale bars in (A–D) = 2 mm. (E) Long bone

length and degree of ossification at E18.5. For each bone, wildtype measurements are shown in the left bar, mutant in the right. The filled portion of each bar depicts the ossified fraction of the bone. Data are averaged independent measurements in each case (N=14 for most measurements, with the exception of wildtype scapula [N = 9], Nipbl+/scapula [N = 13], and *Nipbl*+/- femur [N = 11]). In mutant animals, long bones are \sim 10% shorter than wildtype, and the degree of ossification is decreased by 5-7%. (F,G) Elbow joints of representative wildtype (F) and mutant (G) embryos at E18.5. The olecranon process (arrow) is longer and more pointed in the mutant. (H-K) Defects in cardiac septum formation. The developing atrial septum primum and septum secundum are readily apparent in wildtype heart at E15.5 [(H), arrow] but reduced in Nipbl+/- embryos at the same age (I). At E17.5, a wellformed atrial septum is apparent in wildtype [(J), arrow], but absent in many mutants (K). Scale bars in (G,I,K) = 1 mm. doi:10.1371/journal.pgen.1000650.g001

followed by death (Figure 2F, inset). By 3 weeks of age, the average weight of surviving mutants was only 40% of wildtype, but after weaning this pattern abruptly changed: mutants (even ones that had already begun to show wasting) underwent rapid catch-up growth (Figure 2F), such that by 9 weeks of age they had reached 65–70% of wildtype weight. These observations suggest that, in addition to being intrinsically small, Nipbl+/- mice may have difficulty with suckling, or may receive inadequate nutrition from milk. Remarkably, the weights of children with CdLS also fall further behind age norms during the first year of life, but show significant catch-up growth later on [11].

Nipbl+/- mice exhibit distinctive craniofacial changes

The distinctive craniofacial features of CdLS, including microbrachycephaly, synophrys, upturned nose, and down-turned lips, play an important role in clinical diagnosis [3,6]. Micro-CT analysis was used to assess whether Nipbl+/- mice also display consistent craniofacial changes. Analysis of the skulls of 63 adult mice showed significantly smaller size (microcephaly) among all mutants (N = 23), as well as a variety of significant shape changes (Figure 3). The latter included foreshortening of the anteriorposterior dimensions of the skull (i.e. brachycephaly) and an upward deflection of the tip of the snout (Figure 3B-3E). The upturned nares (Figure 3C and 3E) reflect reduced size of the ethmoid and sphenoid bones, which produces a sunken midface. Together, these shape changes in the basicranium and face are consistent with a greater reduction in the size of chondrocranial, as opposed to dermatocranial, elements within the skull. In addition, an 8% average decrease in bone thickness was also observed (ANOVA, df = 47, F = 18.6, p < 0.01).

Neurological and physiological abnormalities of postnatal *Nipbl*+/- mice

Neurological abnormalities in CdLS include mental retardation, abnormal sensitivity to pain, and seizures [41]. Although Nipbl+/-mice have not been subjected to intensive long-term neurological or behavioral tests, several distinctive behaviors were observed: Repetitive circling (Videos S1, S2, S3) was noted in 20% (34/173; 15 females and 19 males) of adult Nipbl+/- mice (>5 weeks of age), across all generations examined (N₀–N₄). Repetitive behaviors—including twirling in place [42]—are common symptoms in children with CdLS. In addition, 30% (4/13; all males) of Nipbl+/- mice were noted to adopt opisthotonic postures in response to administration of a normal anesthetic dose of avertin (see Materials and Methods), strongly suggesting seizure activity. Seizures are also common in individuals with CdLS [43,44].

We also observed that 15% of *Nipbl*+/- adult mice (24/158; 11 females and 13 males) displayed reflexive hindlimb clasping when



Figure 2. Growth of *Nipbl+/-* mice. (A,B) Photographs of pairs of newborn (A) and 4 week old (B) wildtype and *Nipbl+/-* littermates, illustrating obvious growth retardation. (C–E) Growth curves for N₀ (C), N₁ (D), and N₂ (E) generations, starting at weaning age. Data are pooled by genotype and sex (N₀ wildtype: 15 males and 7 females; N₀ *Nipbl+/-* : 12 males and 7 females; N₁ wildtype 39 males and 28 females; N₁ *Nipbl+/-* : 23 males and 17 females; N₂ wildtype: 72 males and 38 females, N₂ *Nipbl+/-* : 39 males and 23 females): filled squares = male, open circles = female; red symbols = *Nipbl+/-*, black symbols = wildtype; error bars = SEM. Mutant mice of both sexes are smaller, and exhibit less weight gain after maturity. (F) Growth from birth through 6 weeks. Wildtype data (black symbols and lines) are pooled by sex: filled squares = male, open circles = female (data from 28 females and 26 males. error bars = SEM). The records of individual *Nipbl+/-* mice (N₂ and N₃ generation) are shown as connected lines without symbols. Red lines represent individual animals that survived at least 6 weeks (dashed = female; solid = male). Blue lines are *Nipbl+/-* mice that died before weaning. Green lines represent additional mice that died before weaning for which genotype could not be established (due to cannibalism or tissue decomposition). The inset magnifies the pre-weaning interval (1 to 22 days). Whereas wildtype mice grow at a nearly linear rate during the first two weeks of life, the data show that growth of most mutant mice arrests between postnatal days 5 and 15, followed, in about a third of cases, by death several days later. Several of the mutants that stopped growing and lost weight after day 12, however, were able to resume rapid weight gain immediately after weaning. doi:10.1371/journal.pgen.1000650.g002



Figure 3. Craniofacial alterations. Morphometric analysis of craniofacial morphology was based on micro-CT scans of 23 mutant and 40 wildtype mice (No and N1 generation; mixed sexes, ages 218-373 days). (A) Representative reconstructions of wildtype and mutant skulls. From left to right, dorsal, lateral, and ventral views are shown. (B) Average shapes for both genotypes, obtained as described by [131], using the entire sample of scans. (C) Shape distance map [131], showing the distribution of shape effects of the mutation. (D) Results of Euclidean Distance Matrix Shape analysis. This analysis compares the entire set of linear distances between landmarks to identify local differences in form. The two groups are significantly different in shape by a Monte Carlo randomization test (p<0.001). Distances shown are those that differ >5% between groups. Red lines are distances that are relatively smaller in the mutant; blue lines are those that are relatively larger. (E) Results of geometric morphometric analysis (Procrustes superimposed) of craniofacial shape. In this form of analysis, the 3D landmarks representing each individual are scaled to remove size and then superimposed using a translation and rotation step. This yields a dataset in which the differences in landmark position reflect differences in shape independently of size. (i) Principal components analysis can be used to visualize the variation in such a dataset, by transforming a set of correlated variables into a new set of uncorrelated ones, each representing successively smaller portions of the total sample variance. (ii) Variation along a principal component can be represented as a deformation of a wireframe drawn using the landmarks. The first principal component (PC1) distinguishes mutants from wildtypes. The two groups are also significantly different in shape by Goodall's F-test and MANOVA (p<0.001). The first principal component for the combined sample captures the shape variation that distinguishes the groups. The left panel indicates dorsal view of the wireframe and the right indicates lateral view. Blue = wildtype, red = Nipbl+/-. doi:10.1371/journal.pgen.1000650.g003

suspended by their tails (Videos S4, S5), whereas only 2% (6/268) littermates showed the same behavior (Table S1). Hindlimb clasping has been observed in several mouse models of neurological disorders, including Rett's syndrome [45–47], mucolipidosis type IV [48], infantile neuroaxonal dystrophy and neurodegeneration with brain iron accumulation [49], and Huntington's disease [50–53].

Histological examination of mutant brains revealed the presence of all major brain structures, grossly normal lamination of the cerebral and cerebellar cortices, but an overall reduction in brain size, consistent with a 25% reduction in endocranial volume observed with micro-CT (Figure 4A, two-tailed T-test, df = 28, T = 5.7 p < 0.01). Absence or reduction in size of the corpus callosum was occasionally observed in *Nipbl+/* – mice (Figure 4B). Obvious patterning defects were noted only in the midline

cerebellum, where lobe IX displayed specific reductions (Figure 4C). Interestingly, midline cerebellar hypoplasia is one of the few consistently-reported changes in brain anatomy in CdLS [54–56].

Children with CdLS display a range of ophthalmological abnormalities including ptosis, microcornea, nasolacrymal duct obstruction, strabismus, blepharitis and conjunctivitis [57–59]. We noted that 22% of *Nipbl+/-* mice exhibited one or more gross ophthalmological abnormalities (Table S1). Most frequently observed was ocular opacification, observed in 14% of animals (Figure 4D); opacities were often evident as early as three weeks of age. In several cases, this condition was associated with marked periorbital inflammation, and progressed to permanent closure of the eyelids (not shown). Histological analysis revealed inflammatory and fibrotic changes within the corneal epithelium and stroma



Figure 4. Neuroanatomical, ophthalmic, and auditory phenotypes. (A) Measurements from micro-CT analysis (Figure 3) reveal a 25% reduction in endocranial volume in Nipbl+/- mice (data are means±SD from 40 wildtype and 23 mutant skulls). (B) NissI-stained coronal sections of adult wildtype (+/+) and Nipbl+/- brains illustrate reduced brain size, but grossly normal neuroanatomy. MHb, medial habenular nucleus; fr, fasiculus retroflexus; mt, mammillothalamic tract; dq, dentate gyrus; cc, corpus callosum. Scale bar: 1 mm. (C) Cerebellar hypoplasia in Nipbl+/- mice. Cresyl violet-stained midsagittal sections through the cerebellum of adult wildtype and Nipbl+/- mice. Mutant cerebella are smaller overall, with a less welldeveloped folium IX (bracket); note the subfolium (arrow in +/+) that was missing in 100% of analyzed mutants (asterisk; N=3 Nipbl+/- and 3 wildtype littermate controls assessed). Reduction in the size of folium I was also commonly observed in mutants (not shown). Scale bar: 500 µm. (D,E) Corneal pathology in Nipbl+/- mice. (D) External view of wildtype and Nipbl+/- eyes, demonstrating central opacity (green arrowhead) and swelling/inflammation in the periorbital area (arrows) in the mutant. (E) Sections through wildtype and Nipbl+/- eves, demonstrating disruption of corneal structure in the mutant, including infiltration of cells into the stroma and loss of epithelium. Ocular opacification was observed in 14% (24/ 173) of post-weaning animals tested. Scale bar: 100 µm. (F,G) Hearing deficits in Nipbl+/- mice. (F) Auditory brainstem evoked response (ABR) records for a pair of wildtype and Nipbl+/- littermates, performed as described [62]. Stacked curves are responses to successive 10 dB increments of a pure-tone stimulus, and display five characteristic peaks of differing latency. In the Nipbl+/- curves, hearing loss is indicated by the much higher response threshold (this is seen in less than half of mutants). (G) Average background-subtracted sizes of Peaks II, III, and IV (normalized to Peak I to correct for experimental variation due to differences in electrode placement) for the 90 dB tone response of 6 wildtype and 6 mutant animals. Mutants show marked depression of peak III (P<0.02, ANOVA), consistent with abnormalities at the level of the auditory nerve or brainstem. doi:10.1371/journal.pgen.1000650.g004

(Figure 4E), consistent with repeated abrasion or injury. Such injury might arise from neglect due to abnormalities in corneal sensation, from abnormal production or composition of tear fluid, or secondary to periorbital inflammation or infection (e.g. blepharitis; cf. Table S1).

Some degree of hearing loss is observed in almost all individuals with CdLS, and this may play a role in the marked speech disability often seen in this syndrome [60,61]. To assess hearing in Nipbl+/- mice, we measured auditory brainstem evoked responses (ABR [62]). Abnormalities were found in the majority of mutant mice examined (Table S1). In a few cases, markedly increased thresholds to stimulation were observed (Figure 4F). More commonly, stimulus thresholds were within normal limits, but the relative intensities of the components of the ABR were altered. In particular, mutant mice displayed a characteristic reduction in the amplitude of the third peak (at about 3 msec following stimulus), a latency consistent with an abnormality in the auditory nerve and/or early brainstem neural pathways (Figure 4G).

Levels of *Nipbl* expression are reduced by only 25–30% in *Nipbl*+/- mice

The $Nipbl^{564}$ gene-trap mutation is expected to produce a truncated message lacking all but the first exon (Figure S1). Therefore, the level of full-length Nipbl mRNA in Nipbl+/- mice should provide an indication of the activity of the wildtype allele. To measure this level, we used an RNase protection assay based on hybridization to sequences found in exons 10 and 11. Total RNA was analyzed from two tissues: adult liver and E17.5 brain, using age-matched littermate controls. As shown in Figure 5, Nipbl levels in mutants, as a percentage of wildtype levels, were 72–82% in adult liver, and ~70% in embryonic brain. When western blotting was used to quantify levels of NIPBL protein in Nipbl+/- embryo fibroblasts (MEFs), a reduction to about 70% of wildtype levels was observed (Figure S2).

The observation that Nipbl+/- mice exhibit only a 25–30% decrease in transcript and protein expression, rather than an expected decrease of 50%, is consistent with Nipbl gene being autoregulatory. An alternative explanation is that the mutant allele is "leaky", i.e. alternative splicing around the gene trap cassette produces some wildtype message. We favor the former explanation because, in both *Drosophila* and man, the evidence indicates that null

mutation of a single allele of *Nipped-B/NIPBL* produces only a 25–30% drop in transcript levels, the same decrease we observe in *Nipbl+/-* mice [31,63,64]. Thus, even if the *Nipbl* allele studied here is not null, it is probably quite close to being so. More importantly, the degree of decrease in *Nipbl* expression in *Nipbl+/-* mice is comparable to that which causes CdLS in man. Overall the data from multiple species strongly argue that pervasive developmental abnormalities result from remarkably small changes in NIPBL levels.

Sister chromatid cohesion is not significantly affected in *Nipbl*+/- mice

There has been one report of precocious sister chromatid separation (PSCS) in cell lines derived from individuals with CdLS [28], which was not seen in a second study [29]. We found no statistically-significant elevation of PSCS in cultured Nipbl+/- MEFs (Figure S3), Nipbl+/- embryonic stem cells (data not shown), or adult B-lymphocytes (Figure S3). These results suggest that cohesion defects in the Nipbl heterozygotes, if present, are very subtle; they are also in accord with findings in *Drosophila*, where PSCS is seen only when both alleles of Nipped-B are mutated [31].

Dysregulation of gene expression in Nipbl+/- mice

To investigate whether heterozygous loss of Nipbl leads to alterations in transcription, we turned to expression profiling of tissues and cells from Nipbl+/- mice. Because such mice display pervasive developmental abnormalities, transcriptome data can be expected to reflect not only the direct consequences of reduced Nipbl function, but also a potentially large number of transcriptional effects that are secondary consequences of abnormal morphology and physiology. In an effort to minimize the detection of such secondary effects, we focused on profiling samples in which frank pathology was not seen, or had yet to develop by the time of profiling.

The samples chosen for analysis were embryonic day 13.5 (E13.5) brain, and cultures of fibroblasts derived from E15.5 embryos (mouse embryo fibroblasts; MEFs). Although mature brain appears to be functionally abnormal in Nipbl+/- mice (see above), at E13.5 it at least appears anatomically normal. Cultured MEFs were chosen because they are established with similar efficiency from both mutant and wildtype embryos; exhibit similar



Figure 5. *Nipbl* **transcript levels are reduced 25–30% in** *Nipbl+*/*–* **mice.** (A) Autoradiograms showing *Nipbl* and *Gapdh* probes, and protected fragments of 226 (representing *Nipbl* exons 11 and parts of exons 10 and 12) and 131 bases (*Gapdh*), respectively. RNA was prepared from livers of two female littermates (N₀ generation; age = 119 days). The minor protected band at ~185 bp (corresponding to the size of exon 11) most likely arises from the presence of unspliced or alternatively spliced mRNA. (B,C) Quantification of *Nipbl/Gapdh* ratios, from autoradiograms such as in (A), for adult female liver [(B); age = 73 days], and E17.5 brain (C). Mice in each panel are littermates. Hatched bars = wildtype, filled bars = *Nipbl+/–*. Error bars = SD for triplicate (B) or quadruplicate (C) measurements. doi:10.1371/journal.pgen.1000650.g005

morphology and growth characteristics in culture; and by virtue of being maintained *ex vivo*, are freed of the secondary influences of any systemic metabolic or circulatory derangements within Nipbl+/- embryos. Transcriptome analysis was performed using Affymetrix microarrays. MEF RNA samples were obtained from 10 mutant and 9 wildtype embryos taken from three litters (19 separate microarrays); brain RNA was analyzed from 10 mutant and 11 wildtype embryos from two litters (21 separate microarrays).

Gene expression changes were detected in both comparisons. In the brain (Table S2), 1285 probe sets, corresponding to 978 genes, displayed statistically significant differences in expression between wildtype and mutant mice (per-probe-set false discovery rate of Q<0.05). By and large, the effects were small: 97.5% of changes were within 1.5-fold of wildtype expression values; >99.6% were within 2-fold. The single largest statistically-significant change was 2.5-fold. Genes encoding products of virtually all structural and functional categories could be found among those affected, with no dramatic enrichment of any particular functional sets (by Gene Set Enrichment Analysis [65]; data not shown).

In cultured *Nipbl+/* – MEFs, 89 probe sets, corresponding to 81 genes (Table S3), displayed statistically-significant (Q<0.05) differences in expression between wildtype and mutant mice. Again, effects were small: 89% of changes were within 1.5-fold of wildtype, and 99% were within 2-fold. The single largest statistically-significant change was 2.1-fold. The lower number of transcriptional changes identified in MEFs versus brain may not be biologically meaningful, as MEFs happened to display a somewhat higher average within-sample variance than E13.5

brain, making it more difficult for small changes to be judged significant.

As with embryonic brain, transcriptional effects in MEFs involved genes that encode a wide variety of proteins. Although automated analyses failed to single out any particular functional class as being highly overrepresented, manual curation revealed significant changes in the expression of a number of genes implicated in adipogenesis (Figure 6A). For example, Cebpb and *Ebf1*—which encode transcriptional factors central to the process of adipocyte differentiation [66-68]-were both down-regulated in Nibbl+/- MEFs, as were Fabb4 and Aab7, well-known adipocyte markers [69,70]. Other genes down-regulated in Nibbl+/- MEFs (Table S3) could also be found, through literature searches, to exhibit expression positively correlated with adipocyte differentiation, including Adm, Lpar1, Osmr, and Ptx3 [69,71,72]. Several additional genes (Amacr, Avpr1a, Il4ra, Prkcdp, S100b) downregulated in Nipbl+/- MEFs can be inferred, from publiclyavailable expression data, to be enriched in pre-adipocytes and/or brown or white adipose tissue [73-75]. Conversely, Lmo7, which is normally down-regulated during late adipogenic differentiation [71], was found to be up-regulated in Nipbl+/- MEFs. Furthermore, we noted that genes such as Cebpa and Cebpd (transcriptional activators of adipocyte differentiation [66,76]), Il6 (a cytokine stimulator of adipocyte differentiation that controls adiposity in man [77,78]) and Socs3 (an intracellular signaling regulator induced by Il6 [79]), were also down-regulated in the MEF samples, but at false-discovery rates slightly too high to permit their inclusion in Table S3 (Q = 0.065, 0.085, 0.075, and 0.17, respectively).



Figure 6. Alterations in adipogenesis in *Nipbl+/-* **mice.** (A) Gene regulatory network underlying adipogenesis. The adipogenic conversion of pre-adipogenic mesenchyme is under the control of a network of cross-regulating transcription factors, notably C/EBP β , C/EBP α , C/ED α , C/ED

Together, these data raise the possibility that Nipbl+/- mice are specifically impaired in adipogenesis. Support for this idea was obtained by weighing intrascapular fat dissected from adult mutant and wildtype littermates [80]. As shown in Figure 6B, both brown and white fat are substantially depleted in Nipbl+/mice. To correct for the fact that mutant mice are generally smaller than their wildtype littermates, we normalized fat measurements to brain weight (which scales with overall body size). As shown in Figure 6C, even by this measure, Nipbl+/- mice displayed a significant, substantial reduction in body fat. As mentioned earlier, lean body habitus is also a characteristic of CdLS.

To investigate whether the reduction in body fat in Nipbl+/mice reflects an intrinsic defect in the differentiation potential of mutant fibroblasts, we studied adipogenic differentiation in vitro. It is known that embryonic fibroblasts can be converted, in large numbers, to adipocytes by treatment with agents such as glucocorticoids, PPAR- γ agonists, isobutylmethylxanthine and insulin, which stimulate the activity of a core network of proadipogenic transcription factors (C/EBPa, C/EBPb, C/EBPb, PPARy; [81,82]). In response to such agents, we observed no significant difference between Nipbl+/- and wildtype MEFs in terms of the number of adipocytes or adipocyte colonies produced (data not shown). However, when we omitted these pharmacological agents, and measured the (much lower) level of spontaneous adipogenic differentiation that occurs in MEF cultures [83], we observed a substantially-lower level in mutant cultures (Figure 6D-6F). The observation that Nipbl+/- MEFs are impaired in spontaneous, but not induced, adipogenesis implies that their primary defect does not lie downstream of the targets of pharmacological inducers.

Shared transcriptional effects across tissues

Of the 80 genes (not counting Nipbl itself) with significant differential expression in Nipbl+/- MEFs (Table S3), 20% (16/ 80) are also found among the 978 genes whose expression was altered in *Nipbl*+/- embryonic brain (Table S2). Using a more stringent false discovery rate cutoff of Q<0.02 for both samples, we find that 23% (9/40) of differentially expressed MEF genes are among the 560 that are differentially expressed in brain. These data suggest that common transcriptional targets exist in the two tissues. Further support for this idea is obtained by correlating fold-increase or -decrease of affected transcripts. In this case a less conservative approach to false discovery is justified (the goal is to estimate overall correlation between samples, not implicate individual genes), so the logfold changes for all probe sets that exhibited differential expression exceeding an arbitrary t-statistic threshold $(t \ge 2)$ in both tissues were plotted against each other (shown in Figure 7). The data are clearly strongly correlated (R = 0.77), suggesting that at least some of the transcriptional effects of Nipbl deficiency are shared across tissues. Among the genes in which expression changes contributed substantially to the correlation are four members of the protocadherin β cluster (*Pcdh17*, Pcdh20, Pcdh21, Pcdh22; all down-regulated), Lpar1 (also downregulated; encoding the lysophosphatidic acid receptor), Vldlr (down-regulated; encoding a receptor involved in both lipid metabolism and cerebral cortical development), and Stag1 (upregulated; encoding SA1, a cohesin component). Interestingly, in Drosophila, inhibition of Nipped-B expression also leads to upregulation of the ortholog of Stag1 [31]. Recently, STAG1 upregulation has also been seen in lymphoblastoid cell lines of individuals with CdLS [64; Table S5].



Figure 7. Gene expression effects are correlated across cell/ tissue types. Comparison of gene expression profiles of E13.5 brain and MEFs (cf. Table S2 and Table S3, respectively) identified 25 probe sets for which differential expression between wildtype and *Nipbl+/–* samples displayed a t-statistic with an absolute value >2 in both brain and MEFs. For each such probe set, the log2-transformed expression change in MEFs was plotted against the log2-transformed expression change in brain. Points are labeled by gene symbol. The observed correlation (correlation coefficient=0.77) indicates that, for those transcripts affected in both mutant brain and mutant MEFs, the directions and magnitudes of the expression changes are often similar. This suggests that such transcripts may be "direct" targets of NIPBL. Note the presence of *Stag1*, which encodes a cohesin subunit. doi:10.1371/journal.pgen.1000650.g007

Position-specific effects of *Nipbl* on protocadherin beta gene expression

Among the most significant changes common to mutant MEF and brain samples were decreases in expression of transcripts from the 22-gene *Pcdhb* (protocadherin beta) cluster on chromosome 18 (Table S2 and Table S3, Figure 7). As shown in Figure 8A, affected transcripts included *Pcdhb7,16,17,19,20,21* and 22, which lie predominantly at the 3' end of the cluster. This observation raised the possibility that the transcriptional effects of *Nipbl* might be related to the physical locations of genes. However, as genes at the 5' end of the *Pcdhb* cluster tend to be expressed at lower levels than those at the 3' end, lower signal-to-noise ratios might have made small changes in expression at the 5' end more difficult to detect.

To resolve this issue, and to provide independent confirmation of microarray data, quantitative RT-PCR was used to measure transcripts levels at multiple locations throughout the *Pcdhb* cluster (Figure 8B). For these experiments, brain mRNA was prepared at a later developmental stage (E17.5, when most *Pcdhb* transcripts are more highly expressed) from 13 independent samples (7 mutant and 6 wildtype embryos). Robust RT-PCR signals were obtained for 14 of 15 transcripts tested (*Pcdhb2,3,4,5,7,8,9,* 10,13,14,16,17,19, and 22; but not *Pcdhb1*). As shown in Figure 8B, the data support the microarray results from the earlier embryonic stage, and indicate that most transcriptional changes in *Nipbl+/*— brain indeed occur preferentially at the 3' end of the cluster (*Pchdb13,14,15,16,17,19,22*). Additionally, they suggest that at least one 5' gene, *Pcdhb2*, may also be affected.



Figure 8. Position-specific effects on *beta protocadherin (Pcdhb)* **expression.** (A) The protocadherin beta (*Pcdhb*) locus consists of 22 tandemly-oriented, single-exon genes distributed over ~250 kb of chromosome 18. The names of genes that displayed significant reductions in expression in microarray analyses of *Nipbl+/-* E13.5 brain, and *Nipbl+/-* MEFs, are circled in red, and green, respectively. (B) Quantitative RT-PCR was used to measure the levels of 14 *Pcdhb* transcripts in RNA from E17.5 wildtype and *Nipbl+/-* brain. Data are averages from 6 wildtype and 7 mutant samples, presented as percent change from wildtype. (C) Sensitivities of gene expression to *Nipbl* level. Quantitative RT-PCR results for the *Pcdhb* transcripts in (B) were correlated with the levels of *Nipbl* in each mutant and wildtype sample to produce a best-fit regression line that estimates the fold-change in *Pcdhb* transcript per fold-change in *Nipbl*. Error bars representing the standard error of this estimate were obtained from the 67% confidence intervals for the slopes of the regression lines (roughly equivalent to one standard deviation; see Figure S4 for details). Sensitivities and error bars were plotted on an abscissa corresponding to the location of the transcriptional start sites of each of the *Pcdhb* genes. The dashed line is a smooth polynomial fit to the data. Note that a sensitivity of unity simply means that a *Pcdhb* transcript level varies linearly with *Nipbl* levels, whereas a sensitivity of 0.2 means it varies with the 1/5th power of *Nipbl* levels (i.e. very weakly). The data imply that sensitivity is high at both ends of the *Pcdhb* cluster, falling to much lower levels in the middle. doi:10.1371/journal.pgen.1000650.q008

A more revealing analysis of the data can be obtained by correlating Pcdhb transcript levels in each tissue sample, regardless of genotype, against Nipbl transcript levels within that sample (i.e. treating *Nipbl* expression as a quantitative trait; Figure 8C, Figure S4). This approach offers greater discriminatory power because Nipbl expression in individual samples varies significantly, even within mutant and wildtype groups, and occasionally overlaps between the two groups. Indeed, the results of the analysis indicate that *Pcdhb* expression correlates strongly with *Nipbl* transcript level, lending support to the view that *Pcdhb* transcription is directly affected by the amount of NIPBL present in cells. In Figure 8C, the results of such correlations for all 13 tested Pcdhb transcripts are summarized by plotting the slopes of regression lines (the sensitivity of each transcript's expression to Nipbl level) against gene location, with error bars reflecting the strength of correlation for each gene. The results strongly suggest a continuum of sensitivity to Nipbl across the entire *Pcdhb* cluster, with genes at both the 5' and 3' ends being the most sensitive, and those in the middle being least affected.

Discussion

The *Nipbl* mutant mouse as a model for Cornelia de Lange Syndrome

We show here that mice heterozygous for a gene-trap mutation upstream of the first coding exon of *Nipbl* displayed many features of human CdLS, including pre- and postnatal growth retardation, cardiac septal defects, delayed bone development, lean body habitus, microbrachycephaly with characteristic craniofacial changes, behavioral disturbances, ophthalmological abnormalities, cerebellar hypoplasia, and hearing deficits (Figures 1–4, Table S1, Videos S1, S2, S3, S4, S5). These phenotypes remained stable through many generations of outcrossing, and occurred in the context of modest (25–35%) reductions in levels of *Nipbl* mRNA in every tissue measured (embryonic brain, MEFs, adult liver). Similarly modest reductions have recently been reported in cell lines derived from individuals with CdLS [63,64].

In some cases, quantitative agreement between the mouse model and CdLS is remarkable, e.g. fall-off and catch-up in growth rates during early postnatal life, the upturned nose. Yet some common features of CdLS are not observed in the mouse model, such as reduction and fusion abnormalities of the upper limb, which is seen in up to 30–50% of children with CdLS (depending on criteria used). The mutant mouse heart also displays only atrial and not ventricular septal defects, whereas both occur at similar frequency in CdLS. Mutant mice also display some pathological features, such as corneal opacities, that are atypical of CdLS (corneal scarring has been noted, however [58]). Furthermore, the frequency of perinatal mortality in CdLS is estimated at about 10% [8], not as high as in the mutant mouse (although this may simply reflect better postnatal care). Despite these differences, it is clear that the Nipbl+/- mouse is an excellent animal model for many features of CdLS, and provides the first experimental verification that Nipbl mutations cause the syndrome. Interestingly, wide variation in the penetrance or severity of phenotypes, a distinctive feature of CdLS, was also observed in mutant mice. Because the mouse line was maintained on an outbred (CD-1) background (given high mortality of heterozygotes, it was not practical to maintain the line on an inbred background), genetic heterogeneity could have accounted for some of the variability.

It is fascinating that the diverse and severe pathology observed in this study is caused by only a 25–35% decrease in the level of *Nipbl* transcripts (Figure 5; also see Table S2 and Table S3). A recent study of a rare familial case of CdLS involving a mutation in the 5'untranslated region of *NIPBL* suggests that a mere 15% decrease in transcript levels is associated with a clinically significant phenotype [63]. Given the extraordinary sensitivity of development to the level of expression of this one gene, it would not be surprising if an unusually high proportion of disease-causing mutations in *Nipbl* occur in regulatory DNA, where they would be difficult to detect. This could help explain why such a large proportion of CdLS mutations (~40%) have yet to be identified [23,84,85].

Transcriptional dysregulation in the Nipbl+/- mouse

Results of the present study support the view that changes in *Nipbl* level have significant, yet modest, effects on transcription throughout the genome. At present, it is impossible to know how many observed gene expression changes (Table S2 and Table S3) are primary—due to direct transcriptional actions of NIBPL—and how many are downstream consequences of gene misregulation. Among the affected MEF transcripts that were noted to participate in adipogenic differentiation, for example, many are transcriptional targets of each other (Figure 6A), raising the possibility that direct actions of NIPBL may be confined to a subset of these.

Among the most likely candidates for direct NIPBL "targets" are those genes that displayed similar expression changes in both cultured MEFs and E13.5 brain (Figure 7). Prominent among these were genes of the protocadherin beta (*Pcdhb*) cluster. Measurements in later-stage brain confirmed that alterations in gene expression occur throughout the *Pcdhb* locus in *Nipbl+/-*mice, but in a manner that is positionally graded across the cluster (Figure 8). Such effects are consistent with a role for NIPBL in the long-range, coordinated regulation of sets of genes.

Additional evidence for this hypothesis can be found in the E13.5 brain expression data: As shown in Table S4, there are at least 13 other examples of small clusters (usually 2-4 genes) of related (paralogous) genes, in which Nipbl+/- mice display similar expression changes in more than one paralog. Two of these are the well-studied β - and α -globin loci [86], in which long-range cisregulatory elements (locus control regions) are known to control and coordinate expression of different transcripts (the globin transcripts in brain RNA presumably come from fetal erythrocytes in the tissue). Interestingly, whereas decreases in expression were seen at all four β -globin genes in *Nipbl*+/- samples, the magnitudes varied greatly among the genes within each cluster (arguing against the trivial possibility that *Nipbl*+/- brains simply have less blood in them). In fact, the single greatest gene expression change in the entire study (~ 2.5 fold decrease) involved one of the transcripts (*Hbb-bh1*) of the β -globin locus.

It is known that the transcriptional insulator protein CTCF plays an important role in establishing chromatin boundary elements at the β -globin locus [87]. The *Pcdhb* locus is also flanked, at least in man, by sites occupied by CTCF [88], the functional significance of which has yet to be studied. CTCF insulation is also involved in control of the *myc*-locus [89], which is highly significantly downregulated in E13.5 *Nipbl+/* – brain (Table S2). Even the *Igf2/H19* locus, at which long-range, CTCF-dependent regulation of gene silencing has been shown to occur [90,91], displayed evidence of *H19* down-regulation (by \sim 20%) in the *Nipbl+/* – brain, albeit at lower statistical significance (Q = 0.14).

In view of recent work showing that cohesin and CTCF binding sites extensively co-localize in the mammalian genome (including at the β -globin, Igf2/H19 and myc loci [35–37]), and that cohesin contributes to CTCF function [35,37], it is reasonable to speculate that at least some of the transcriptional effects in Nipbl+/- tissues arise from impaired CTCF function. It should be noted, however, that CTCF sites are far more common (>13,000 per mammalian genome) than Nipbl-sensitive genes, and we find no clear correlation between the locations of Nipbl-sensitive genes (or the magnitudes of transcriptional effects in those genes) and known or predicted CTCF-sites (X. Xie and N. Infante, personal communication). It remains possible that only a subset of NIPBL transcriptional effects is related to CTCF function. Indeed, it is possible that NIPBL acts primarily by influencing other aspects of long-range cis-regulatory interaction (e.g. histone methylation, DNA looping), which simply take place frequently at CTCF-regulated loci.

Despite the extensive overlap between the phenotypes of Nipbl+/- mice and CdLS, it is interesting to note that the gene expression changes recently reported in lymphoblastoid cell lines of CdLS individuals [64] exhibit only limited overlap (6–8%; cf. Table S5) with those we observed in mouse embryo fibroblasts and embryonic brain (Table S2 and Table S3). So far it is unclear whether this stems from a high degree of tissue-specificity in the expression of genes that are directly affected by NIPBL; a high proportion of indirect NIPBL targets (which might be more likely to vary from tissue to tissue); or the effects of differences in genomic organization between mouse and man.

Phenotypic significance of transcriptional changes in the *Nipbl*+/- mouse

Among the gene expression changes detected in *Nipbl+/-*MEFs and brain (Table S2 and Table S3) one can find many genes that, when mutated in mice or man, produce phenotypes that overlap with CdLS. These include skeletal and craniofacial abnormalities (*Lpar1*, *Pitx2*, *Satb2*, *Tcof1*, *Trps1*); heart defects (*Adm*, *Cited2*, *Cxcl12*, *Gja1*, *Hey2*, *Pitx2*, *Mef2c*); reduced body size (*Ebf1*, *Lpar1*, *Hsd3b7*, *Mef2c*); decreased adiposity (*Cebpb*, *Ebf1*, *Lpar1*, *Npy*, *Vldlr*); behavioral abnormalities (*Avpr1a*, *Ctnnd2*, *Lpar1*, *Vldlr*); seizures (*Cdk5r1*, *Gabrb1*, *Gabrb2*, *Neto1*, *Nr4a3*, *Plcb1*, *S100b*, *Sv2b*); and hearing deficits (*Cldn11*, *Eya*, *Gjb2*, *Gjb6*) [66,92–124].

For most of the genes mentioned above, however, phenotypes are observed only with complete loss of gene function. In the few cases in which significant heterozygous phenotypes are seen (e.g. Satb2, Pitx2, Trps1, Tcof1, Cited2), expression changes in the same genes in Nipbl+/- samples are not even as great as would be expected for heterozygous loss. Of course, it is possible that greater expression changes occur at other stages, or in other tissues, than those sampled here. However, the alternative interpretation is that phenotypes in *Nipbl*+/- mice, and in individuals with CdLS, arise from the collective effects of small changes in the expression of many genes. Although we are not yet in a position to distinguish between these hypotheses, we recognize that this issue is closely related to a major unanswered question in human genetics: whether most common disease phenotypes arise from large effects at a few loci, or from very many loci of small effect. The results of the present study suggest that further study of CdLS, and related "cohesinopathies" [125,126], could shed light on a fundamental question of widespread importance.

Materials and Methods

Ethics Statement: All animals were handled in strict accordance with good animal practice as defined by the relevant national and/ or local animal welfare bodies, and all animal work was approved by the University of California Irvine Institutional Animal Care and Use Committee (protocol 1998-1656).

A search for Nipbl sequences in mouse gene-trap databases (http://www.genetrap.org/) initially identified two targeted ES cell lines (generated using the E14 parental cell line, which has a 129/Ola background). One of these (RRS564) contains a genetrap in the intron between exon1 and exon 2; the other (RRJ102) in intron 25 (the exon numbering of [16] for the human gene is used here). Gene-trap constructs are designed to terminate transcription and translation, producing a truncated or absent protein product. Both cell lines were injected into blastocysts of C57BL/6 mice (for the RRJ102 cell line, 83 blastocysts were injected; for RRS564, 324 blastocysts were injected). Multiple male chimeras were obtained and bred against outbred (CD-1; Charles River) females. Germ line progeny (distinguishable by chinchilla coat color) were obtained only from RRS564-derived chimeras, and further work on RRJ102 was suspended. The RRS564 allele is hereafter referred to as $Nipbl^{564}$ and mice heterozygous for this allele as Nipbl+/- for simplicity. Nipbl+/mice were maintained under normal laboratory conditions, and the line propagated by successive rounds of outcrossing to CD-1 mice. Offspring were genotyped using LacZ-(Forward 5'-TGAT-GAAAGCTGGCTACAG-3' and Reverse 5'-ACCACCGCAC-GATAGAGATT-3') primers.

Anatomical and histological evaluations were performed using fresh-frozen or paraformaldehyde fixed tissues. In some cases, fixation was carried out by cardiac perfusion. Alcian Blue/Alizarin Red staining was carried out as described [127]. Hematoxylineosin and cresyl violet staining was carried out using standard techniques. Micro-CT analysis of adult (>90 days) skulls was performed using a Scanco VivaCT as described [128,129]. Craniofacial shape was assessed using geometric morphometric techniques, and cranial vault thickness was assessed in 3D as described [130,131]. Scapular fat pads were dissected and measured as described [80]. Auditory brainstem response recordings were generated as described [62]. Briefly, a cohort of young adult mutant and littermate control animals were anesthetized with avertin (2.5% solution of tribromoethanol in tert-amyl alcohol; 20 µl/g body weight administered by i.p. injection), and subcutaneous electrodes inserted at the level of the brainstem to record neural potentials evoked by a variety of clicks and tones introduced into one ear.

Embryo fibroblasts (MEFs) were cultured from E15.5 *Nipbl+/*and wildtype littermate embryos as described [127]. Metaphase spreads of MEFs were prepared from cells cultured for 12 hrs in medium supplemented with 0.1 μ g/ml colchicine. Trypsinized cells were pelleted, incubated in 75 mM KCl for 20 min at 37°C, then re-pelleted and fixed in 3:1 methanol/acetic acid. Cells were dropped onto glass slides and stained with 4'-6-Diamidino-2phenylindole or Giemsa. Microscopic assessment was carried out for each slide by three independent observers who were blinded to the genotypes of the sample.

B cells were isolated from mouse spleens by immunomagnetic depletion with anti-CD43 beads (Miltenyi Biotech), cultured in RPMI1640 with 10% fetal bovine serum, and stimulated with lipopolysaccharide (25 μ g/ml; Sigma) and IL4 (5 ng/ml; Sigma) for 3 days. Cells were arrested at mitosis by treatment with 0.1 μ g/ml colcemid (Roche) for 1 hour, and metaphase chromosome spreads prepared following standard procedures. Cells were

stained with 4'-6-Diamidino-2-phenylindole and images of metaphases acquired with an Axioplan2 upright microscope (Zeiss), using Metamorph software.

Measurements of *Nipbl* levels by RNase protection were made according to standard methods. The *Nipbl* probe contained 39 bases of exon 10, all 183 bases of exon 11, and 4 bases of exon 12. There is no expressed sequence tag evidence supporting alternative splicing of exons 10–11, and in situ hybridization studies in mouse embryos indicated they are ubiquitously expressed, so it was felt that this probe would provide a good indication of overall levels of expression. Briefly, for each reaction, 20 µg of total RNA was hybridized with ³²P-labeled probes for *Nipbl* (90,000 cpm) and *Gapdh* (glyceraldehyde 3-phosphate dehydrogenase; containing 116 bases of exon 4 and 15 bases of exon 5; 20,000 cpm) and processed according to manufacturer's instructions (Ambion RPA III kit). Samples were run on a 5% polyacrylamide/8M urea gel, dried, and bands quantified by phosphorimager.

For measurement of NIPBL protein levels, MEFs were lysed in cold buffer containing 50 mM Tris (pH 7.4), 150 mM NaCl, 1% Triton X-100, 1% sodium deoxycholate, 0.1% SDS, and protease inhibitors [1 mM phenylmethylsulfonyl fluoride (PMSF), 1 mM EDTA, 2 μ g/ml aprotinin, 2 μ g/ml leupeptin, and 2 μ g/ml pepstatin]. Total protein (20 μ g) from 3 wildtype MEF homogenates and 3 Nipbl+/- MEF homogenates was separated, in duplicate, on 7.5% SDS-PAGE gels and transferred onto Immobilon-P membranes (Millipore, Bedford, MA). The membranes were blocked with 2% BSA and sequentially incubated with the anti-GAPDH (1:200,000, 6C5; Ambion, Austin, TX) and anti-NIPBL-N (1:60000, anti-NIPBL antibody was produced in rabbit from a GST fusion protein containing amino acids 1-380 of human NIPBL, and was affinity purified using original antigen). The membranes were then incubated with the horseradish peroxidase-conjugated anti-rabbit antibody (1:10000) and detected using chemiluminescence. Images were scanned and densitometry performed.

To evaluate spontaneous adipogenic differentiation, MEFs (passage 2) were seeded into 96-well plates at 7,500 cells/well, maintained in Dulbecco's modified Eagle's Medium with 10% fetal bovine serum, 100 U/ml penicillin and 100 µg/ml streptomycin at 37°C, cultured to confluence, and maintained for 7 additional days. Lipid accumulation was visualized by staining with Oil Red O. Briefly, cells were washed with PBS and fixed with 10% formaldehyde (30 minutes at room temperature), rinsed, and permeabilized in 60% isopropanol for 2-5 minutes. Isopropanol was removed and Oil Red O solution (Chemicon, Temecula, CA) was added for 30 minutes. Wells were washed several times with PBS then counterstained with Hoechst 33258 (0.01 mg/ml). Four wildtype and five Nipbl+/- MEF lines were analyzed. Each was plated in triplicate wells and 7 fields (at $10 \times$ magnification) within each well were photographed and lipidcontaining cells (stained lipids appear red under phase contrast) and nuclei (appear blue with fluorescent microscopy) were counted using Image J software (NIH) to detect nuclei (nucleus counter plug-in: approximately 1000-1200 nuclei were detected per field) and lipid-containing cells (point picker plug-in).

Analysis of gene expression was performed on total RNA isolated using the trizol method from MEFs, adult liver, or manually dissected E13.5 brain. Hybridization and data collection were carried out by the Broad Institute (Cambridge, MA). RNA was labeled and hybridized to Affymetrix Murine 430A 2.0 (for MEFs) or 430 2.0 (for brain) array chips using the protocol described at http://www.broad.mit.edu/mpr/publications/projects/ Leukemia/protocol.html, and data were analyzed using GenePattern

software (http://www.broad.mit.edu/cancer/software/genepattern). Expression data sets were assembled from individual CEL files using the RMA algorithm with quantile normalization. Data were log2-transformed and transcripts with near-background expression filtered. Measures of statistical significance were obtained by permutation testing [132], using the Comparative Marker Selection module. Significance is presented in terms of per-sample false discovery rates, or Q-values [133]. Data were also analyzed using D-chip software (http://www.biostat.harvard.edu/complab/dchip), which yielded similar enrichment sets (not shown). Probe sets were annotated, and gene locations obtained, according to the NCBI m37 mouse assembly, and Affymetrix annotation files ver. 28 (March 2009).

For measurements of transcript abundance by quantitative PCR, RNA (5 μ g) from E17.5 mouse brains was reverse transcribed with Superscript II, oligo dT, and random hexamers according to manufacturer's instructions (Invitrogen, Carlsbad, CA). Reactions were assembled using iQ SYBR Green Supermix (BioRad, Hercules, CA) and processed in 20 μ l volumes with 1 μ l of cDNA (diluted 1:25) and primers at a final concentration of 100 nM. Specificity of amplification was verified for each reaction by examination of the corresponding melt curve. Normalization was carried out using beta-2 microglobulin as a standard, and genomic amplification controlled for using samples prepared without reverse transcription. All PCR reactions were 95°C for 4 min and then 40 cycles of 95°C 10 sec, 61°C 30 sec and 72°C 30 sec. Primers were:

Pcdhb2: agcccacctggtagatgttg and attggggatgattggtttca; Pcdhb3: cctggaaatacaccgcagaa and cctagacatggacccagcaa; Pcdhb4: cagtcagteceaaceteca and tgaactgtggtcateceagae; Pedhb5: cagaggggaaatcaggaaca and gggcttaaactggcaatgaa; Pcdhb7: accccacacaggaagttgag and ctttatccccacgaaaagca; Pcdhb8: gccttggcttctgtgtcttc and caccactgacatccaccaag; Pcdhb9: atgcctggtgaacactttcc and gcagtggggactttccataa; *Pcdhb10*: gctgaccctcacctctcttg and accaccacgagtaccaaagc; Pcdhb13: ggcttctctcagccctacc and cagcaccacagacaagagga; Pcdhb14: cattgcacataggcaccatc and tgatggagatgagcgagttg; Pcdhb16: tggcttctetcagecetace and aacageageacagaeaceag; Pedhb17: geaagteetggctttctttg and ggatatctctgccaggtcca; Pcdhb19: gacaaggcaagtcctgcttc and ccccaggtcctttaccaaat; Pcdhb22: tatcatcgctcaccaatcca and cagagctccatctgtcacca, beta-2-microglobulin :, atgggaagccgaacatactg and cagtetcagtgggggggaat, and Nipbl (exons 6-7): agtecatatgccccacagag and accggcaacaataggacttg. PCR product sizes were between 107 and 182 bp.

Supporting Information

Figure S1 The *Nipbl* gene trap allele. (A) The *Nipbl* gene consists of 47 exons, distributed over 150 kbp on Chromosome 15. The *Nipbl*⁵⁶⁴ allele inserts a β-geo gene trap cassette within intron 1. (B) *Nipbl* exon 1 forward primer and β-geo reverse primer [arrowheads in (A)] detect *Nipbl*-β-geo fusion mRNA in reverse-transcribed RNA from *Nipbl*⁵⁶⁴ ES cells (lane 1). No signal is detected in wildtype ES cells (lane 2). The structure of the fusion RNA was confirmed by sequencing. Lane M = 100 bp DNA ladder.

Found at: doi:10.1371/journal.pgen.1000650.s001 (0.24 MB TIF)

Figure S2 NIPBL protein levels in E15.5 *Nipbl+/-* MEFs. Immunoblot comparing levels of NIPBL protein in wildtype (+/+) and *Nipbl+/-* MEFs. Protein extracts from MEF cultures established from 2 wildtype and 2 *Nipbl+/-* mice were subjected to electrophoresis, blotted, and probed with an antibody directed against the N-terminus of the human NIPBL protein (see Materials and Methods). Immunoblotting for Gapdh served as a protein loading control. Densitometric analysis revealed that NIPBL protein was reduced by 27% in *Nipbl+/-* MEFs. Found at: doi:10.1371/journal.pgen.1000650.s002 (0.16 MB TIF)

Figure S3 Assessment of sister chromatid cohesion in Nipbl+/cells. (A) Metaphase spreads of cultured MEFs were scored according to which of the following characterized the greatest proportion of chromatids: separated (completely separated with no connection at the centromere); loosened (connected only at the centromere); and closed (no gap; chromatids remain connected). Data are expressed as mean \pm S.E.M. for \geq 100 metaphase spreads prepared from 2 independent lines of MEFs of each genotype. No significant differences in category frequency were observed between genotypes. (B) Metaphase spreads of cultured Blymphocytes were analyzed for sister chromatid cohesion as in (A). Of 89 wildtype and 147 Nipbl+/- metaphase spreads, from spleen cells of two wildtype and three Nipbl+/- animals, no examples of loosened or separated chromatids were seen. Representative examples of wildtype (+/+) and Nipbl+/- images are shown.

Found at: doi:10.1371/journal.pgen.1000650.s003 (0.55 MB TIF)

Figure S4 Position-specific effects on beta protocadherin (*Pcdhb*) expression. Quantitative PCR was used to measure levels of *Nipbl* and *Pcdhb* transcripts in RNA samples of E17.5 brain from 6 wildtype and 7 *Nipbl+/* – mice. Normalized crossing point threshold values (Δ Ct) for each *Pcdhb* transcript were plotted against Δ Ct values for *Nipbl* (wildtype = black; mutant = red), and linear regression lines were calculated along with 95% confidence intervals (dashed curves) around those lines. Panels show data for *Pcdhb2*, *3*, *4*, *5*, *7*, *8*, *9*, *10*, *13*, *14*, *16*, *17*, *19 and 22*. The slope of each regression line corresponds to the input-output sensitivity [134], i.e. the fold change in *Pcdhb* transcript for any given fold change in *Nipbl*. r = correlation coefficient.

Found at: doi:10.1371/journal.pgen.1000650.s004 (8.40 MB EPS)

Table S1 Structural phenotypes in Nipbl+/- mice, and comparable clinical findings in CdLS. Mouse data are presented as percent of affected animals for each genotype (Wildtype, Nipbl+/-) and the number of mice assessed (N). Data are pooled for five generations (N₀-N₄). The incidence of comparable clinical findings in CdLS is listed in the right-hand columns.

Found at: doi:10.1371/journal.pgen.1000650.s005 (0.07 MB PDF)

Table S2 Gene expression changes in embryonic day 13.5 Nipbl+/- brain. Fold changes and per-probe-set false discovery rates (Q-values) were calculated as described (see Materials and Methods). Data were obtained from 10 mutant and 11 wildtype samples. Abs(Score) = the absolute value of the t-test statistic for each sample. Q-value = per-entry false discovery rate. Among the 1182 probe sets with Q<0.05, there are 978 independent genes. Among the 751 probe sets with Q<0.02, there are 560 independent genes.

Found at: doi:10.1371/journal.pgen.1000650.s006 (0.27 MB XLS)

Table S3 Gene expression changes in *Nipbl+/-* MEFs. Fold changes and per-probe-set false discovery rates (Q-values) were calculated as described (see Materials and Methods). Data were obtained from 10 mutant and 9 wildtype samples. Abs(Score) = the absolute value of the t-test statistic for each sample. Q-value = per-entry false discovery rate. Among the 89 probe sets with Q<0.05, there are 81 independent genes. Among the 43 probe sets with Q<0.02, there are 40 independent genes.

Found at: doi:10.1371/journal.pgen.1000650.s007 (0.03 MB XLS)

Table S4 Paralog clusters that display altered expression in the E13.5 Nipbl+/- brain. Expression data from E13.5 brain were

analyzed to identify clustered groups of related genes that were affected in the Nipbl+/- mutant. In addition to the 22-gene protocadherin beta locus (see Figure 7), there were 13 cases of two-to-four paralog clusters exhibiting significant transcriptional changes. Data are shown for all the paralogs in each cluster, and for any interspersed non-paralogous genes lying within such clusters, for which data were available (i.e. informative probe sets existed, and expression was above background). In all but one case, gene expression changes among genes within a cluster were always in the same direction. Two of the clusters (the alpha- and beta-globin loci) are well-known locus control regions (see text).

Found at: doi:10.1371/journal.pgen.1000650.s008 (0.04 MB PDF)

Table S5 Genes displaying expression changes in both *Nipbl+/*mice (this study) and CdLS lymphoblastoid cell lines [64]. Gene symbols refer to human loci. Where the standard symbols for mouse orthologs are not identical, the mouse symbol is given in parentheses. Found at: doi:10.1371/journal.pgen.1000650.s009 (0.05 MB PDF)

Video S1 Circling behavior of Nipbl+/- mice. Videos S1, S2, and S3 contrast the repetitive daytime circling behavior displayed by two sibling Nipbl+/- mice (Videos S1 and S2) with those of a wildtype littermate (N₁ generation, Video S3). Circling behavior was consistently more pronounced for the animal in this video than in the one in Video S2.

Found at: doi:10.1371/journal.pgen.1000650.s010 (6.07 MB MOV)

Video S2 Circling behavior of Nipbl+/- mice. Videos S1, S2, and S3 contrast the repetitive daytime circling behavior displayed by two sibling Nipbl+/- mice (Videos S1 and S2) with those of a wildtype littermate (N₁ generation, Video S3).

Found at: doi:10.1371/journal.pgen.1000650.s011 (5.81 MB MOV)

Video S3 Behavior of a wildtype littermate. Videos S1, S2, and S3 contrast the repetitive daytime circling behavior displayed by

References

- de Lange C (1933) Sur un type nouveau de degeneration (typus Amstelodamensis). Arch Med Enfants 36: 713–719.
- Jackson L, Kline AD, Barr MA, Koch S (1993) de Lange syndrome: a clinical review of 310 individuals. Am J Med Genet 47: 940–946.
- Ireland M, Donnai D, Burn J (1993) Brachmann-de Lange syndrome. Delineation of the clinical phenotype. Am J Med Genet 47: 959–964.
- Braddock SR, Lachman RS, Stoppenhagen CC, Carey JC, Ireland M, et al. (1993) Radiological features in Brachmann-de Lange syndrome. Am J Med Genet 47: 1006–1013.
- Mehta AV, Ambalavanan SK (1997) Occurrence of congenital heart disease in children with Brachmann-de Lange syndrome. Am J Med Genet 71: 434–435.
- Allanson JE, Hennekam RC, Ireland M (1997) De Lange syndrome: subjective and objective comparison of the classical and mild phenotypes. J Med Genet 34: 645–650.
- Tsukahara M, Okamoto N, Ohashi H, Kuwajima K, Kondo I, et al. (1998) Brachmann-de Lange syndrome and congenital heart disease. Am J Med Genet 75: 441–442.
- Barisic I, Tokic V, Loane M, Bianchi F, Calzolari E, et al. (2008) Descriptive epidemiology of Cornelia de Lange syndrome in Europe. Am J Med Genet A 146A: 51–59.
- Lachman R, Funamura J, Szalay G (1981) Gastrointestinal abnormalities in the Cornelia de Lange syndrome. Mt Sinai J Med 48: 236–240.
- Maruiwa M, Nakamura Y, Motomura K, Murakami T, Kojiro M, et al. (1988) Cornelia de Lange syndrome associated with Wilms' tumour and infantile haemangioendothelioma of the liver: report of two autopsy cases. Virchows Arch A Pathol Anat Histopathol 413: 463–468.
- Kline AD, Barr M, Jackson LG (1993) Growth manifestations in the Brachmann-de Lange syndrome. Am J Med Genet 47: 1042–1049.
- Kline AD, Stanley C, Belevich J, Brodsky K, Barr M, et al. (1993) Developmental data on individuals with the Brachmann-de Lange syndrome. Am J Med Genet 47: 1053–1058.
- Moss JF, Oliver C, Berg K, Kaur G, Jephcott L, et al. (2008) Prevalence of autism spectrum phenomenology in Cornelia de Lange and Cri du Chat syndromes. Am J Ment Retard 113: 278–291.

two sibling Nipbl+/- mice (Videos S1 and S2) with those of a wildtype littermate (N₁ generation), shown in this video.

Found at: doi:10.1371/journal.pgen.1000650.s012 (5.08 MB MOV)

Video S4 Limb-clasping behavior. Videos S4 and S5 contrast the limb-clasping behavior (especially of hindlimbs) displayed by 15% of *Nipbl+/-* mutants when suspended by their tails (this video) with the limb-spreading displayed by nearly all wildtype mice (Video S5).

Found at: doi:10.1371/journal.pgen.1000650.s013 (1.27 MB AVI)

Video S5 Behavior of a wildtype littermate. Videos S4 and S5 contrast the limb-clasping behavior (especially of hindlimbs) displayed by 15% of Nipbl+/- mutants when suspended by their tails (Video S4) with the limb-spreading displayed by nearly all wildtype mice (this video).

Found at: doi:10.1371/journal.pgen.1000650.s014 (0.89 MB AVI)

Acknowledgments

The authors are grateful to Aice Quach, Annie Espinosa, Aileen Alvarez, and Theresa Chua for assistance with mouse husbandry and phenotypic analysis; to Tom Fielder and the University of California Irvine Transgenic Mouse Facility for microinjection services for the generation of NipbH/- chimeric mice; to Todd Golub, Oleg Iartchouk, and David Peck of the Broad Institute for assistance with microarray experiments and analysis; to Peter J. Gruber for assistance with embryonic cardiac anatomy; and to Dale Dorsett and Ian Krantz for helpful discussions. ADL and ALC dedicate this paper to the memory of Isabel Eva Calof Lander.

Author Contributions

Conceived and designed the experiments: SK ALC MSL JAD LK BH ADL. Performed the experiments: SK ALC RS MELB CMY MPH TL JAD AN LK BH ADL. Analyzed the data: SK ALC RS MELB MPH AC LK BH ADL. Contributed reagents/materials/analysis tools: KY. Wrote the paper: SK ALC ADL.

- Luzzani S, Macchini F, Valade A, Milani D, Selicorni A (2003) Gastroesophageal reflux and Cornelia de Lange syndrome: typical and atypical symptoms. Am J Med Genet A 119: 283–287.
- Opitz JM (1985) The Brachmann-de Lange syndrome. Am J Med Genet 22: 89–102.
- Krantz ID, McCallum J, DeScipio C, Kaur M, Gillis LA, et al. (2004) Cornelia de Lange syndrome is caused by mutations in NIPBL, the human homolog of Drosophila melanogaster Nipped-B. Nat Genet 36: 631–635.
- Tonkin ET, Wang TJ, Lisgo S, Bamshad MJ, Strachan T (2004) NIPBL, encoding a homolog of fungal Scc2-type sister chromatid cohesion proteins and fly Nipped-B, is mutated in Cornelia de Lange syndrome. Nat Genet 36: 636–641.
- Dorsett D, Krantz ID (2009) On the molecular etiology of Cornelia de Lange syndrome. Ann N Y Acad Sci 1151: 22–37.
- Gillis LA, McCallum J, Kaur M, DeScipio C, Yaeger D, et al. (2004) NIPBL mutational analysis in 120 individuals with Cornelia de Lange syndrome and evaluation of genotype-phenotype correlations. Am J Hum Genet 75: 610–623.
- Dorsett D (2004) Adherin: key to the cohesin ring and Cornelia de Lange syndrome. Curr Biol 14: R834–R836.
- Hagstrom KA, Meyer BJ (2003) Condensin and cohesin: more than chromosome compactor and glue. Nat Rev Genet 4: 520–534.
- Musio A, Selicorni A, Focarelli ML, Gervasini C, Milani D, et al. (2006) Xlinked Cornelia de Lange syndrome owing to SMC1L1 mutations. Nat Genet 38: 528–530.
- Deardorff MA, Kaur M, Yaeger D, Rampuria A, Korolev S, et al. (2007) Mutations in cohesin complex members SMC3 and SMC1A cause a mild variant of cornelia de Lange syndrome with predominant mental retardation. Am J Hum Genet 80: 485–494.
- Vega H, Waisfisz Q, Gordillo M, Sakai N, Yanagihara I, et al. (2005) Roberts syndrome is caused by mutations in ESCO2, a human homolog of yeast ECO1 that is essential for the establishment of sister chromatid cohesion. Nat Genet 37: 468–470.
- Schule B, Oviedo A, Johnston K, Pai S, Francke U (2005) Inactivating mutations in ESCO2 cause SC phocomelia and Roberts syndrome: no phenotype-genotype correlation. Am J Hum Genet 77: 1117–1128.

- Zhang B, Jain S, Song H, Fu M, Heuckeroth RO, et al. (2007) Mice lacking sister chromatid cohesion protein PDS5B exhibit developmental abnormalities reminiscent of Cornelia de Lange syndrome. Development 134: 3191–3201.
- Zhang B, Chang J, Fu M, Huang J, Kashyap R, et al. (2009) Dosage effects of cohesin regulatory factor PDS5 on mammalian development: implications for cohesinopathies. PLoS One 4: e5232. doi:10.1371/journal.pone.0005232.
- Kaur M, DeScipio C, McCallum J, Yaeger D, Devoto M, et al. (2005) Precocious sister chromatid separation (PSCS) in Cornelia de Lange syndrome. Am J Med Genet A 138: 27–31.
- Vrouwe MG, Elghalbzouri-Maghrani E, Meijers M, Schouten P, Godthelp BC, et al. (2007) Increased DNA damage sensitivity of Cornelia de Lange syndrome cells: evidence for impaired recombinational repair. Hum Mol Genet 16: 1478–1487.
- Rollins RA, Morcillo P, Dorsett D (1999) Nipped-B, a Drosophila homologue of chromosomal adherins, participates in activation by remote enhancers in the cut and Ultrabithorax genes. Genetics 152: 577–593.
- Rollins RA, Korom M, Aulner N, Martens A, Dorsett D (2004) Drosophila nipped-B protein supports sister chromatid cohesion and opposes the stromalin/Scc3 cohesion factor to facilitate long-range activation of the cut gene. Mol Cell Biol 24: 3100–3111.
- Dorsett D, Eissenberg JC, Misulovin Z, Martens A, Redding B, et al. (2005) Effects of sister chromatid cohesion proteins on cut expression during wing development in Drosophila. Development 132: 4743–4753.
- Dorsett D (2007) Roles of the sister chromatid cohesion apparatus in gene expression, development, and human syndromes. Chromosoma 116: 1–13.
- Misulovin Z, Schwartz YB, Li XY, Kahn TG, Gause M, et al. (2008) Association of cohesin and Nipped-B with transcriptionally active regions of the Drosophila melanogaster genome. Chromosoma 117: 89–102.
- Parelho V, Hadjur S, Spivakov M, Leleu M, Sauer S, et al. (2008) Cohesins functionally associate with CTCF on mammalian chromosome arms. Cell 132: 422–433.
- Stedman W, Kang H, Lin S, Kissil JL, Bartolomei MS, et al. (2008) Cohesins localize with CTCF at the KSHV latency control region and at cellular c-myc and H19/Igf2 insulators. Embo J 27: 654–666.
- Wendt KS, Yoshida K, Itoh T, Bando M, Koch B, et al. (2008) Cohesin mediates transcriptional insulation by CCCTC-binding factor. Nature 451: 796–801.
- Jahnke P, Xu W, Wulling M, Albrecht M, Gabriel H, et al. (2008) The Cohesin loading factor NIPBL recruits histone deacetylases to mediate local chromatin modifications. Nucleic Acids Res 36: 6450–6458.
- Froster UG, Gortner L (1993) Thrombocytopenia in the Brachmann-de Lange syndrome. Am J Med Genet 46: 730–731.
- Roposch A, Bhaskar AR, Lee F, Adedapo S, Mousny M, et al. (2004) Orthopaedic manifestations of Brachmann-de Lange syndrome: a report of 34 patients. J Pediatr Orthop B 13: 118–122.
- Hawley PP, Jackson LG, Kurnit DM (1985) Sixty-four patients with Brachmann-de Lange syndrome: a survey. Am J Med Genet 20: 453–459.
- Johnson HG, Ekman P, Friesen W (1976) A behavioral phenotype in the de Lange syndrome. Pediatr Res 10: 843–850.
- Berney TP, Ireland M, Burn J (1999) Behavioural phenotype of Cornelia de Lange syndrome. Arch Dis Child 81: 333–336.
- 44. Kline AD, Krantz ID, Sommer A, Kliewer M, Jackson LG, et al. (2007) Cornelia de Lange syndrome: clinical review, diagnostic and scoring systems, and anticipatory guidance. Am J Med Genet A 143A: 1287–1296.
- Chang Q, Khare G, Dani V, Nelson S, Jaenisch R (2006) The disease progression of Mecp2 mutant mice is affected by the level of BDNF expression. Neuron 49: 341–348.
- Gemelli T, Berton O, Nelson ED, Perrotti LI, Jaenisch R, et al. (2006) Postnatal loss of methyl-CpG binding protein 2 in the forebrain is sufficient to mediate behavioral aspects of Rett syndrome in mice. Biol Psychiatry 59: 468–476.
- Guy J, Hendrich B, Holmes M, Martin JE, Bird A (2001) A mouse Mecp2-null mutation causes neurological symptoms that mimic Rett syndrome. Nat Genet 27: 322–326.
- Venugopal B, Browning MF, Curcio-Morelli C, Varro A, Michaud N, et al. (2007) Neurologic, gastric, and opthalmologic pathologies in a murine model of mucolipidosis type IV. Am J Hum Genet 81: 1070–1083.
- Shinzawa K, Sumi H, Ikawa M, Matsuoka Y, Okabe M, et al. (2008) Neuroaxonal dystrophy caused by group VIA phospholipase A2 deficiency in mice: a model of human neurodegenerative disease. J Neurosci 28: 2212–2220.
- Auerbach W, Hurlbert MS, Hilditch-Maguire P, Wadghiri YZ, Wheeler VC, et al. (2001) The HD mutation causes progressive lethal neurological disease in mice expressing reduced levels of huntingtin. Hum Mol Genet 10: 2515–2523.
- Baquet ZC, Gorski JA, Jones KR (2004) Early striatal dendrite deficits followed by neuron loss with advanced age in the absence of anterograde cortical brainderived neurotrophic factor. J Neurosci 24: 4250–4258.
- Dragatsis I, Levine MS, Zeitlin S (2000) Inactivation of Hdh in the brain and testis results in progressive neurodegeneration and sterility in mice. Nat Genet 26: 300–306.
- Reddy PH, Williams M, Charles V, Garrett L, Pike-Buchanan L, et al. (1998) Behavioural abnormalities and selective neuronal loss in HD transgenic mice expressing mutated full-length HD cDNA. Nat Genet 20: 198–202.

- Hayashi M, Sakamoto K, Kurata K, Nagata J, Satoh J, et al. (1996) Septo-optic dysplasia with cerebellar hypoplasia in Cornelia de Lange syndrome. Acta Neuropathol (Berl) 92: 625–630.
- Ozkinay F, Cogulu O, Gunduz C, Levent E, Ozkinay C (1998) A case of Brachman de Lange syndrome with cerebellar vermis hypoplasia. Clin Dysmorphol 7: 303–305.
- Yamaguchi K, Ishitobi F (1999) Brain dysgenesis in Cornelia de Lange syndrome. Clin Neuropathol 18: 99–105.
- Levin AV, Seidman DJ, Nelson LB, Jackson LG (1990) Ophthalmologic findings in the Cornelia de Lange syndrome. J Pediatr Ophthalmol Strabismus 27: 94–102.
- Nallasamy S, Kherani F, Yaeger D, McCallum J, Kaur M, et al. (2006) Ophthalmologic findings in Cornelia de Lange syndrome: a genotypephenotype correlation study. Arch Ophthalmol 124: 552–557.
- Wygnanski-Jaffe T, Shin J, Perruzza E, Abdolell M, Jackson LG, et al. (2005) Ophthalmologic findings in the Cornelia de Lange Syndrome. J Aapos 9: 407–415.
- Sataloff RT, Spiegel JR, Hawkshaw M, Epstein JM, Jackson L (1990) Cornelia de Lange syndrome. Otolaryngologic manifestations. Arch Otolaryngol Head Neck Surg 116: 1044–1046.
- Sakai Y, Watanabe T, Kaga K (2002) Auditory brainstem responses and usefulness of hearing aids in hearing impaired children with Cornelia de Lange syndrome. Int J Pediatr Otorhinolaryngol 66: 63–69.
- Zheng QY, Johnson KR, Erway LC (1999) Assessment of hearing in 80 inbred strains of mice by ABR threshold analyses. Hear Res 130: 94–107.
- Borck G, Zarhrate M, Cluzeau C, Bal E, Bonnefont JP, et al. (2006) Father-todaughter transmission of Cornelia de Lange syndrome caused by a mutation in the 5' untranslated region of the NIPBL Gene. Hum Mutat 27: 731–735.
- Liu J, Zhang Z, Bando M, Itoh T, Deardorff MA, et al. (2009) Transcriptional dysregulation in *MIPBL* and cohesin mutant human cells. PLoS Biol 7: e1000119. doi:10.1371/journal.pbio.1000119.
- Subramanian A, Tamayo P, Mootha VK, Mukherjee S, Ebert BL, et al. (2005) Gene set enrichment analysis: a knowledge-based approach for interpreting genome-wide expression profiles. Proc Natl Acad Sci U S A 102: 15545–15550.
- Tanaka T, Yoshida N, Kishimoto T, Akira S (1997) Defective adipocyte differentiation in mice lacking the C/EBPbeta and/or C/EBPdelta gene. EMBO J 16: 7432–7443.
- Rosen ED, MacDougald OA (2006) Adipocyte differentiation from the inside out. Nat Rev Mol Cell Biol 7: 885–896.
- Jimenez MA, Akerblad P, Sigvardsson M, Rosen ED (2007) Critical role for Ebf1 and Ebf2 in the adipogenic transcriptional cascade. Mol Cell Biol 27: 743–757.
- Bujalska IJ, Quinkler M, Tomlinson JW, Montague CT, Smith DM, et al. (2006) Expression profiling of 11beta-hydroxysteroid dehydrogenase type-1 and glucocorticoid-target genes in subcutaneous and omental human preadipocytes. J Mol Endocrinol 37: 327–340.
- Maeda N, Funahashi T, Hibuse T, Nagasawa A, Kishida K, et al. (2004) Adaptation to fasting by glycerol transport through aquaporin 7 in adipose tissue. Proc Natl Acad Sci U S A 101: 17801–17806.
- Burton G, Nagarajan R, Peterson C, McGehee Jr R (2004) Microarray analysis of differentiation-specific gene expression during 3T3-L1 adipogenesis. Gene 329: 167–185.
- Abderrahim-Ferkoune A, Bezy O, Chiellini C, Maffei M, Grimaldi P, et al. (2003) Characterization of the long pentraxin PTX3 as a TNFalpha-induced secreted protein of adipose cells. J Lipid Res 44: 994–1000.
- Su AI, Wiltshire T, Batalov S, Lapp H, Ching KA, et al. (2004) A gene atlas of the mouse and human protein-encoding transcriptomes. Proc Natl Acad Sci U S A 101: 6062–6067.
- Hua K, Deng J, Harp JB (2004) Interleukin-4 inhibits platelet-derived growth factor-induced preadipocyte proliferation. Cytokine 25: 61–67.
- Gesta S, Bluher M, Yamamoto Y, Norris AW, Berndt J, et al. (2006) Evidence for a role of developmental genes in the origin of obesity and body fat distribution. Proc Natl Acad Sci U S A 103: 6676–6681.
- Nerlov C (2008) C/EBPs: recipients of extracellular signals through proteome modulation. Curr Opin Cell Biol 20: 180–185.
- Strandberg L, Mellstrom D, Ljunggren O, Grundberg E, Karlsson MK, et al. (2008) IL6 and IL1B polymorphisms are associated with fat mass in older men: the MrOS Study Sweden. Obesity (Silver Spring) 16: 710–713.
- Qi L, Zhang C, van Dam RM, Hu FB (2007) Interleukin-6 genetic variability and adiposity: associations in two prospective cohorts and systematic review in 26,944 individuals. J Clin Endocrinol Metab 92: 3618–3625.
- Croker BA, Krebs DL, Zhang JG, Wormald S, Willson TA, et al. (2003) SOCS3 negatively regulates IL-6 signaling in vivo. Nat Immunol 4: 540–545.
- 80. Ailhaud G(2001) Adipose Tissue Protocols. Totowa (New Jersey): Humana Press. 334 p.
- Gregoire FM, Smas CM, Sul HS (1998) Understanding adipocyte differentiation. Physiol Rev 78: 783–809.
- Rosen ED, Walkey CJ, Puigserver P, Spiegelman BM (2000) Transcriptional regulation of adipogenesis. Genes Dev 14: 1293–1307.
- Chen PL, Riley DJ, Chen Y, Lee WH (1996) Retinoblastoma protein positively regulates terminal adipocyte differentiation through direct interaction with C/ EBPs. Genes Dev 10: 2794–2804.

- Borck G, Zarhrate M, Bonnefont JP, Munnich A, Cormier-Daire V, et al. (2007) Incidence and clinical features of X-linked Cornelia de Lange syndrome due to SMC1L1 mutations. Hum Mutat 28: 205–206.
- Schoumans J, Wincent J, Barbaro M, Djureinovic T, Maguire P, et al. (2007) Comprehensive mutational analysis of a cohort of Swedish Cornelia de Lange syndrome patients. Eur J Hum Genet 15: 143–149.
- Miele A, Dekker J (2008) Long-range chromosomal interactions and gene regulation. Mol Biosyst 4: 1046–1057.
- Tanimoto K, Sugiura A, Omori A, Felsenfeld G, Engel JD, et al. (2003) Human beta-globin locus control region HS5 contains CTCF- and developmental stage-dependent enhancer-blocking activity in erythroid cells. Mol Cell Biol 23: 8946–8952.
- Kim TH, Abdullaev ZK, Smith AD, Ching KA, Loukinov DI, et al. (2007) Analysis of the vertebrate insulator protein CTCF-binding sites in the human genome. Cell 128: 1231–1245.
- Ohlsson R, Renkawitz R, Lobanenkov V (2001) CTCF is a uniquely versatile transcription regulator linked to epigenetics and disease. Trends Genet 17: 520–527.
- Hark AT, Schoenherr CJ, Katz DJ, Ingram RS, Levorse JM, et al. (2000) CTCF mediates methylation-sensitive enhancer-blocking activity at the H19/ Igf2 locus. Nature 405: 486–489.
- 91. Kurukuti S, Tiwari VK, Tavoosidana G, Pugacheva E, Murrell A, et al. (2006) CTCF binding at the H19 imprinting control region mediates maternally inherited higher-order chromatin conformation to restrict enhancer access to Igf2. Proc Natl Acad Sci U S A 103: 10684–10689.
- Lin H, Grosschedl R (1995) Failure of B-cell differentiation in mice lacking the transcription factor EBF. Nature 376: 263–267.
- Caron KM, Smithies O (2001) Extreme hydrops fetalis and cardiovascular abnormalities in mice lacking a functional Adrenomedullin gene. Proc Natl Acad Sci U S A 98: 615–619.
- Dyck RH, Bogoch II, Marks A, Melvin NR, Teskey GC (2002) Enhanced epileptogenesis in S100B knockout mice. Brain Res Mol Brain Res 106: 22–29.
- Bielsky IF, Hu SB, Szegda KL, Westphal H, Young IJ (2004) Profound impairment in social recognition and reduction in anxiety-like behavior in vasopressin V1a receptor knockout mice. Neuropsychopharmacology 29: 483–493.
- Contos JJ, Fukushima N, Weiner JA, Kaushal D, Chun J (2000) Requirement for the lpA1 lysophosphatidic acid receptor gene in normal suckling behavior. Proc Natl Acad Sci U S A 97: 13384–13389.
- Kitamura K, Miura H, Miyagawa-Tomita S, Yanazawa M, Katoh-Fukui Y, et al. (1999) Mouse Pitx2 deficiency leads to anomalies of the ventral body wall, heart, extra- and periocular mesoderm and right pulmonary isomerism. Development 126: 5749–5758.
- Liu W, Selever J, Lu MF, Martin JF (2003) Genetic dissection of Pitx2 in craniofacial development uncovers new functions in branchial arch morphogenesis, late aspects of tooth morphogenesis and cell migration. Development 130: 6375–6385.
- Teubner B, Michel V, Pesch J, Lautermann J, Cohen-Salmon M, et al. (2003) Connexin30 (Gjb6)-deficiency causes severe hearing impairment and lack of endocochlear potential. Hum Mol Genet 12: 13–21.
- Israely I, Costa RM, Xie CW, Silva AJ, Kosik KS, et al. (2004) Deletion of the neuron-specific protein delta-catenin leads to severe cognitive and synaptic dysfunction. Curr Biol 14: 1657–1663.
- Weninger WJ, Floro KL, Bennett MB, Withington SL, Preis JI, et al. (2005) Cited2 is required both for heart morphogenesis and establishment of the leftright axis in mouse development. Development 132: 1337–1348.
- Ng D, Pitcher GM, Szilard RK, Sertie A, Kanisek M, et al. (2009) Neto1 is a novel CUB-domain NMDA receptor-interacting protein required for synaptic plasticity and learning. PLoS Biol 7: e41. doi:10.1371/journal.pbio.1000041.
- 103. MacDonald ST, Bamforth SD, Chen CM, Farthing CR, Franklyn A, et al. (2008) Epiblastic Cited2 deficiency results in cardiac phenotypic heterogeneity and provides a mechanism for haploinsufficiency. Cardiovasc Res 79: 448–457.
- 104. Dixon J, Jones NC, Sandell LL, Jayasinghe SM, Crane J, et al. (2006) Tcof1/ Treacle is required for neural crest cell formation and proliferation deficiencies that cause craniofacial abnormalities. Proc Natl Acad Sci U S A 103: 13403–13408.
- 105. Liu S, Liu F, Schneider AE, St Amand T, Epstein JA, et al. (2006) Distinct cardiac malformations caused by absence of connexin 43 in the neural crest and in the non-crest neural tube. Development 133: 2063–2073.
- Gassmann M, Shaban H, Vigot R, Sansig G, Haller C, et al. (2004) Redistribution of GABAB(1) protein and atypical GABAB responses in GABAB(2)-deficient mice. J Neurosci 24: 6086–6097.
- 107. Schuler V, Luscher C, Blanchet C, Klix N, Sansig G, et al. (2001) Epilepsy, hyperalgesia, impaired memory, and loss of pre- and postsynaptic GABA(B) responses in mice lacking GABA(B(1)). Neuron 31: 47–58.
- Xu T, Bajjalich SM (2001) SV2 modulates the size of the readily releasable pool of secretory vesicles. Nat Cell Biol 3: 691–698.

- Kim D, Jun KS, Lee SB, Kang NG, Min DS, et al. (1997) Phospholipase C isozymes selectively couple to specific neurotransmitter receptors. Nature 389: 290–293.
- Chae T, Kwon YT, Bronson R, Dikkes P, Li E, et al. (1997) Mice lacking p35, a neuronal specific activator of Cdk5, display cortical lamination defects, seizures, and adult lethality. Neuron 18: 29–42.
- Depreux FF, Darrow K, Conner DA, Eavey RD, Liberman MC, et al. (2008) Eya4-deficient mice are a model for heritable otitis media. J Clin Invest 118: 651–658.
- Ste Marie L, Luquet S, Cole TB, Palmiter RD (2005) Modulation of neuropeptide Y expression in adult mice does not affect feeding. Proc Natl Acad Sci U S A 102: 18632–18637.
- Vong LH, Ragusa MJ, Schwarz JJ (2005) Generation of conditional Mcf2cloxP/loxP mice for temporal- and tissue-specific analyses. Genesis 43: 43–48.
- 114. Fischer A, Klamt B, Schumacher N, Glaeser C, Hansmann I, et al. (2004) Phenotypic variability in Hey2 -/- mice and absence of HEY2 mutations in patients with congenital heart defects or Alagille syndrome. Mamm Genome 15: 711-716.
- 115. Malik TH, Von Stechow D, Bronson RT, Shivdasani RA (2002) Deletion of the GATA domain of TRPS1 causes an absence of facial hair and provides new insights into the bone disorder in inherited tricho-rhino-phalangeal syndromes. Mol Cell Biol 22: 8592–8600.
- Weeber EJ, Beffert U, Jones C, Christian JM, Forster E, et al. (2002) Reclin and ApoE receptors cooperate to enhance hippocampal synaptic plasticity and learning. J Biol Chem 277: 39944–39952.
- 117. Cohen-Salmon M, Ott T, Michel V, Hardelin JP, Perfettini I, et al. (2002) Targeted ablation of connexin26 in the inner ear epithelial gap junction network causes hearing impairment and cell death. Curr Biol 12: 1106–1111.
- Nagasawa T, Hirota S, Tachibana K, Takakura N, Nishikawa S, et al. (1996) Defects of B-cell lymphopoiesis and bone-marrow myelopoiesis in mice lacking the CXC chemokine PBSF/SDF-1. Nature 382: 635–638.
- Frykman PK, Brown MS, Yamamoto T, Goldstein JL, Herz J (1995) Normal plasma lipoproteins and fertility in gene-targeted mice homozygous for a disruption in the gene encoding very low density lipoprotein receptor. Proc Natl Acad Sci U S A 92: 8453–8457.
- Hesslein DG, Fretz JA, Xi Y, Nelson T, Zhou S, et al. (2009) Ebf1-dependent control of the osteoblast and adipocyte lineages. Bone 44: 537–546.
- 121. Fernandez AP, Serrano J, Tessarollo L, Cuttitta F, Martinez A (2008) Lack of adrenomedullin in the mouse brain results in behavioral changes, anxiety, and lower survival under stress conditions. Proc Natl Acad Sci U S A 105: 12581–12586.
- 122. Shea HC, Head DD, Setchell KD, Russell DW (2007) Analysis of HSD3B7 knockout mice reveals that a 3alpha-hydroxyl stereochemistry is required for bile acid function. Proc Natl Acad Sci U S A 104: 11526–11533.
- 123. Britanova O, Depew MJ, Schwark M, Thomas BL, Miletich I, et al. (2006) Satb2 haploinsufficiency phenocopies 2q32-q33 deletions, whereas loss suggests a fundamental role in the coordination of jaw development. Am J Hum Genet 79: 668–678.
- 124. Gow A, Davies C, Southwood CM, Frolenkov G, Chrustowski M, et al. (2004) Deafness in Claudin 11-null mice reveals the critical contribution of basal cell tight junctions to stria vascularis function. J Neurosci 24: 7051–7062.
- Liu J, Krantz ID (2008) Cohesin and human disease. Annu Rev Genomics Hum Genet 9: 303–320.
- McNairn AJ, Gerton JL (2008) Cohesinopathies: One ring, many obligations. Mutat Res 647: 103–111.
- 127. Nagy A, Gertsenstein M, Vintersten K, Behringer R (2002) Manipulating the Mouse Embryo: A Laboratory Manual. New York: Cold Spring Harbor Laboratory Press. 800 p.
- Hallgrimsson B, Willmore K, Dorval C, Cooper DM (2004) Craniofacial variability and modularity in macaques and mice. J Exp Zoolog B Mol Dev Evol 302: 207–225.
- Hallgrimsson B, Brown JJ, Ford-Hutchinson AF, Sheets HD, Zelditch ML, et al. (2006) The brachymorph mouse and the developmental-genetic basis for canalization and morphological integration. Evol Dev 8: 61–73.
- Hildebrand T, Ruegsegger P (1997) A new method for the model-independent assessment of thickness in three-dimensional images. J Microsocopy 185: 67–75.
- Kristensen E, Parsons TE, Hallgrimsson B, Boyd SK (2008) A novel 3-D image-based morphological method for phenotypic analysis. IEEE Trans Biomed Eng 55: 2826–2831.
- Golub TR, Slonim DK, Tamayo P, Huard C, Gaasenbeek M, et al. (1999) Molecular classification of cancer: class discovery and class prediction by gene expression monitoring. Science 286: 531–537.
- Storey JD, Tibshirani R (2003) Statistical significance for genomewide studies. Proc Natl Acad Sci U S A 100: 9440–9445.
- Reeves GT, Fraser SE (2009) Biological systems from an engineer's point of view. PLoS Biol 7: e21. doi: 10.1371/journal.pbio.1000021.